

Electronic Theses and Dissertations, 2004-2019

2007

Change In The Indian Accounting Profession: Three Studies Related To The Entry Of The Big Four Accounting Firms In India

Vikram G. Desai University of Central Florida



Find similar works at: https://stars.library.ucf.edu/etd University of Central Florida Libraries http://library.ucf.edu

This Doctoral Dissertation (Open Access) is brought to you for free and open access by STARS. It has been accepted for inclusion in Electronic Theses and Dissertations, 2004-2019 by an authorized administrator of STARS. For more information, please contact STARS@ucf.edu.

STARS Citation

Desai, Vikram G., "Change In The Indian Accounting Profession: Three Studies Related To The Entry Of The Big Four Accounting Firms In India" (2007). *Electronic Theses and Dissertations, 2004-2019.* 3139. https://stars.library.ucf.edu/etd/3139



CHANGE IN THE INDIAN ACCOUNTING PROFESSION: THREE STUDIES RELATED TO THE ENTRY OF THE BIG FOUR ACCOUNTING FIRMS IN INDIA

by

VIKRAM DESAI

Chartered Accountant, The Institute of Chartered Accountants of India, 1994 M.B.A, University of Arkansas, 1998

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in the Kenneth G. Dixon School of Accounting in the College of Business Administration at the University of Central Florida Orlando, Florida

Fall Term 2007

Major Professor: Robin W. Roberts



ABSTRACT

This dissertation focuses on the globalization of audit markets. In particular, this dissertation is studying the entry of the Big Four accounting firms into India post-economic and political reforms of the early 1990s. The dissertation is comprised of three separate, but related studies. Each study appeals to prior research in accounting and related disciplines to examine the entry of the Big Four accounting firms in India. The first study appeals to audit market and economic research on incumbent pricing to examine ways in which local accounting firms in India adapted to the competition introduced by the Big Four accounting firms. The second study is an account of the change in the organizational field of the Indian accounting profession caused by the entry of the multinational accounting firms from 1990 to 2005 from a social constructionist perspective using the model of nonisomorphic change. The third study examines the change in the Indian accounting profession from 1990 to 2005 caused by the entry of the Big Four accounting firms in India from a critical perspective. It appeals to the theories of globalization to examine the change. Taken together, these studies attempt to provide the Big Four accounting firms useful information about the pricing strategies likely to be faced by them from local accounting firms in a new market, provide insights into the multiple roles played by professional associations in the process of radical change in the organizational field, and emphasize that globalization of accounting markets has not been accompanied by a level playing field for the local accounting profession in the globalized markets.



ACKNOWLEDGMENTS

There are not enough words to describe the help and support provided by my dissertation chair and friend, Robin Roberts in completing this dissertation. I can say without any hesitation that it is due to his guidance that I was able to reach at this stage in my PhD endeavor. I am also thankful to my committee members Peggy Dwyer, Gregory Trompeter, and Stephen Sivo for their insightful comments to constantly improve this dissertation.



TABLE OF CONTENTS

LIST OF FIGURES	V i
LIST OF TABLES	vi
CHAPTER ONE: GENERAL INTRODUCTION	8
Study One: Incumbent Pricing Response: Evidence of the entry of the Big Four accounting firms in India	
Study Two: Impacts of Globalization: Evidence from the Indian accounting profession	8
Study Three: Entry of the Big Four accounting firms in India: An Institutional Perspective	e 9
Overall Contribution	10
CHAPTER TWO: STUDY ONE: Incumbent pricing respones: Evidence of the entry of the Four accounting firms in India	_
Introduction	11
Background Indian Accounting Profession and Entry of the Big Four accounting firms in India Entry Pricing Literature Incumbent pricing expectation in an audit setting	15
Hypotheses and Research Methodology	24 25 rms
Sample Selection and Model Specification	
Results Preliminary Analysis: Estimation of Big Four premium	34
Discussions and Conclusions	40
References	43
CHAPTER THREE: Study two: Impacts of globalization: evidence from the indian Accou	
Introduction	47
Theoretical Context Theoretical perspectives on globalization Theoretical Model: Institutional model of globalization	50
Methods	
Site and Data Sources The Indian Accounting Profession	
THE INGRAL (ACCOUNTIES FIVICAMENT)	



The case of India: Entry of multinational accounting firms	61
The Foreign Exchange crisis in 1990: Liberalization of the Indian Economy	61
Post Liberalization: The Entry Process	64
Post Entry: Entry in the traditional auditing and assurance market	70
Support of the Indian Government and the Indian Financial Institutions	75
Theoretical Insights	81
Concluding Remarks	85
References	89
CHAPTER FOUR: study three: entry of the Multinational accounting firms in INDIA: an	
institutional perspective	92
Introduction	92
Theoretical framework	95
Organizational field	95
The Neo Institutional Model of Change	96
Methods	99
Sites and Data Sources	99
The Indian Accounting Profession	. 102
The Case Study	104
Precipitating Jolts and Deinstitutionalization: The Foreign Exchange Crisis (1990)	
Preinstitutionalization: Entry of the Big Four accounting firms in the Indian Market	110
Theorization and Diffusion: The Vision Project (1998)	114
Discussion and Conclusion	125
Role of professional associations in the process of change	125
Theorization and Diffusion	128
References	. 131
appendix a: List of authoratative sources	138
Appendix B: Interview details	140



LIST OF FIGURES

Figure 1:Neo-Institutional-Model-of-change	127
Figure Tineo-institutional-wodel-of-change	



LIST OF TABLES

Table 1. Size distribution of CA firms as at 01/04/2001	16
Table 2. List of Surrogate Accounting firms	17
Table 3. Empirical findings of incumbent pricing responses to entry	20
Table 4. Models and Hypotheses	33
Table 5. Preliminary Analysis: Estimation of Big Four premium	35
Table 6. Descriptive statistics and within-client correlation matrix	37
Table 7. Incumbent Pricing- Audit fee estimation from 1996-2005	39
Table 8. Timeline of events in the process of globalization of the Indian accounting profession.	.55
Table 9. Timeline of events for the ICAI.	58
Table 10. Size distribution of CA firms as at 01/04/2001	60
Table 11. List of Disinvestments with Foreign and Indian advisors	78



CHAPTER ONE: GENERAL INTRODUCTION

This dissertation focuses on the globalization of audit markets. In particular, this dissertation is studying the entry of the Big Four accounting firms into India post-economic and political reforms of the early 1990s. The dissertation is comprised of three separate, but related studies. Each study appeals to prior research in accounting and related disciplines to examine the entry of the Big Four accounting firms in India. Each study is discussed separately in the following three sub-sections.

Study One:

Incumbent Pricing Response: Evidence of the entry of the Big Four accounting firms in India

The first study appeals to audit market and economic research on incumbent pricing to examine ways in which local accounting firms in India adapted to the competition introduced by the Big Four accounting firms. Using data from the annual reports of Indian companies from 1995 to 2005, the study examines whether factors such as the age of the accounting firm, market structure, and the tenure of the auditor are associated with incumbent accounting firms' pricing response to the entry of the Big Four accounting firms. Results may suggest newer incumbents will cut prices more than older incumbents, medium sized incumbents cut prices more than small sized incumbents, while incumbents with longer tenure cut prices less following entry.

Study Two:

Impacts of Globalization: Evidence from the Indian accounting profession

The third study examines the change in the Indian accounting profession from 1990 to 2005 caused by the entry of the Big Four accounting firms in India from a critical perspective. It appeals to the theories of globalization to examine the change. The study attempts to enforce the view that globalization is not a natural and inevitable process, but that it is a politically



constructed institution that is driven by powerful international bodies such as multinational accounting firms and the International Monetary Fund, using instruments of coercion and pressure. Specifically, I intend to explore the interconnectedness of professional and state politics with global forces by examining the role played by professional associations and the state in the process of globalization. The results will show how the multinational accounting firms, well positioned in the international economic system, are capable of not only changing the boundaries of the local accounting profession, but also becoming competitive forces in a relatively short span of time from 1990 to 2005.

Study Three:

Entry of the Big Four accounting firms in India: An Institutional Perspective

The second study is an account of the change in the organizational field of the Indian accounting profession caused by the entry of the multinational accounting firms from 1990 to 2005 from a social constructionist perspective using the model of nonisomorphic change. The change was precipitated by the foreign exchange crisis that almost turned the country bankrupt in 1991. To deal with the shortage of foreign exchange reserves leading to non payment of external debt, the government of India requested a Stand-By Arrangement from the International Monetary Fund in August 1991. One of the stipulations of the Stand-By Arrangement was that the multinational accounting firms, essentially the Big Four accounting firms, be allowed to provide consulting services in India. Results may reveal that professional associations are not only concerned with preserving the existing prescriptions but also play an important role in legitimating the change through a process of discourse in the field. This discourse enables the change to be resisted, debated, and eventually accepted.



Overall Contribution

Taken together, these studies attempt to achieve the following broad objectives:

- Provide the Big Four accounting firms useful information about the pricing strategies likely to be faced by them from local accounting firms in a new market.
- 2. Determine the effect of the entry of the Big Four accounting firms in a new market on the price structure of audit services in the audit market.
- 3. Provide insights into the multiple roles played by professional associations in the process of radical change in the organizational field.
- 4. Reinforce the theory of institutional globalization, which states that transnational institutional forces work proactively to create global markets to the detriment of the weak nation states.
- 5. Emphasize that globalization of accounting markets has not been accompanied by a level playing field for the local accounting profession in the globalized markets.

The remainder of this dissertation presents each of the three studies in detail.



CHAPTER TWO: STUDY ONE: INCUMBENT PRICING RESPONES: EVIDENCE OF THE ENTRY OF THE BIG FOUR ACCOUNTING FIRMS IN INDIA

Introduction

The purpose of this study is to examine the effect of the entry of the Big Four accounting firms on the audit fees charged by the local accounting firms in India. The entry of new firms has important effects on incumbent firms and consumers. New entrants increase competition, reducing market share and profits of incumbent firms. It is obvious, therefore, that incumbents have strong incentives to deter entry, while entry is beneficial to consumers as new firms also introduce new products and processes, forcing incumbents to become more efficient and innovative (Geroski,1995). In the event of entry, incumbents have to decide on the most effective response. They may reduce prices before or after entry to deter potential entrants or simply to maximize current profits in the face of increased competition (Simon, 2005). Alternatively, they may leave prices unchanged and use other strategies to prevent entry such as increasing advertising expenditures(Cubbin and Domberger, 1988) or R&D and patent protection (Smiley, 1988). Specifically, this study uses audit fees and other relevant data to examine the incumbent accounting firms¹ pricing responses to the increasing presence of the Big Four accounting firms in the Indian audit market.

This study is interesting for several reasons. First, extant research in the audit literature has little to offer on the pricing responses of incumbents to the entry of Big Four accounting firms in any market, mainly because in the western countries such as USA, UK, or Australia, the Big Four accounting firms are the oldest firms in the market. India is a unique market to study

¹ Incumbent firms in this setting are the local accounting firms and therefore the terms are used interchangeably throughout the paper



because, before 1995, the presence of the Big Four accounting firms in India was negligible (The New Indian Express, 2002. The Chartered Accountants Action Committee For Level Playing Field (CAAC) published a White Paper regarding the influence of multinational accounting firms in India, especially the Big Four accounting firms and the response of the Indian accounting profession. Prior to 1949, accounting was controlled and regulated by the Government, which eventually vested the regulatory power in the Institute of Chartered Accountants of India through an enactment in the parliament in 1949. Since that time, the Indian economy was comprised of small and medium sized companies and the accounting profession was consequently also dominated by small and medium sized firms. The market for audit services, therefore, was very local and competitive, due to the absence of any one large firm dominating the market. The number of medium sized firms (No of partners >10) was only 52 out of the total of 42,339 firms operating in the Indian audit market in the year ended 2001(CAAC, 2002). Local accounting firms did not face competition from multinational accounting firms and were not required to adopt strategies to maintain their dominance over the practice of accountancy. Since 1995, the market shares of the Big Four accounting firms have increased substantially (CAAC, 2002). In 2004, the Big Four accounting firms accounted for about 62 % of the audit fees in the Indian audit market (Prime Academy, 2006)². Therefore, interest exists in examining the response of the local accounting firms to the increasing presence of the Big Four accounting firms in the Indian audit market.

Second, theories of entry deterrence and incumbent response to entry have generated a variety of predictions. Early limit pricing studies predicted that incumbents would reduce prices before entry, but not change prices after entry (Bain 1956, 1976; Modigliani, 1958; Sylos-Labini,

² It should be noted that Prime Academy derived this market share from publicly available data of about 1394 listed Indian companies as of March 31, 2004.

1962). However, game-theoretic studies predicted that incumbents may reduce prices as a way to drive out entrants and deter future entrants (Milgrom and Roberts, 1982a). These theoretical studies prompted empirical research regarding incumbent responses to entry, producing inconsistent results. Some studies find that entry has a negative effect on incumbent prices (Frank and Salkever, 1997); others find that entry has no effect on incumbent pricing (Yamawacki, 2002). There is little explanation for these inconsistent findings in the entry literature. The present study seeks to address these inconsistencies by analyzing the factors which might influence incumbents' pricing responses and providing useful insights into the pricing strategies used by accounting firms in the face of competition from new entrants.

Simon (2005) was the first study to offer an explanation for why some firms respond to entry more aggressively than others by examining incumbent pricing response in the magazine market. He suggested that incumbents vary in their incentive to respond to entry and those incumbents with greater incentives to respond are more likely to respond aggressively. The results indicated that incumbent and market characteristics moderate the incumbents' incentive to respond to entry; newer incumbents cut prices more than older incumbents, while incumbents that compete in fewer and in more competitive markets cut prices less following entry. However, as Simon himself points out, the results in his study could have been biased due to the high rate of entry in the magazine industry and therefore calls for similar studies in other industries with lower entry rates. The audit market in India had a very low entry rate due to the competitive nature of the market prior to the entry of the Big Four accounting firms and also due to entry barriers imposed by legislation (CAAC, 2002). In this paper, I intend to test the predictions of the entry pricing literature in the Indian audit market by studying the pricing response of the local accounting firms in India to the entry of the Big Four accounting firms.



Finally, Geroski (1995) in his survey of the empirical work on entry, points out that the "short-run effects of entry are likely to be much less than the long-run effects" (pp.436). The reason is that it takes the new entrants a period of five or ten years to reach a competitive par with incumbents. Most of the empirical work he points out has been directed at measuring short-run rather than long-run effects associated with entry, and, as a result, it is not surprising that the entry literature has yielded such inconsistent results. I intend to address this issue by studying the response of the local accounting firms in India for a period of ten years starting from the year of entry (1996-2005).

While this study examines only the pricing responses of the local accounting firms to the entry of the Big Four accounting firms in the Indian audit market, it should be stressed that incumbent accounting firms can adopt alternative non-pricing responses to the threat of entry (Geroski, 1995). This study examines pricing behavior because theory offers testable predictions regarding heterogeneous pricing responses by incumbents and because, prices are observable and measurable, while many non-price responses are unobserved or difficult to measure such as increased scope of services or cost cutting measures adopted by firms (Simon, 2005).

This study has important implications for the Big Four accounting firms as it will provide them with useful information about the pricing strategies likely to be faced by them from local accounting firms in a new market. Local accounting firms in emerging markets can also gain useful insights about the pricing strategies adopted by the Big Four accounting firms when they enter a market. For instance, they may be able to learn whether the Big Four accounting firms offer substantial discounts when they enter a market or whether they maintain the premiums charged by them in the established markets. This study may also throw light on the effect of the entry of the Big Four accounting firms in an emerging economy on the price structure of audit



services in the audit market. For instance, the entry of the Big Four accounting firms might inflate the audit fee structure prevalent in an existing market or might lead to reduced audit fees due to increased competition.

The remainder of the paper is structured as follows. The next section provides a brief background of the Indian accounting profession and the entry of the Big Four firms in the Indian market. Further, it describes the entry pricing literature and incumbent pricing expectations in an audit setting. Section III presents the hypothesis and research design, followed by a description of the sample and the regression models used to test the hypothesis in Section IV. Section V discusses the results and the final section summarizes the study and its principal conclusions.

Background

Indian Accounting Profession and Entry of the Big Four accounting firms in India

In India, prior to 1949, the profession of accountancy was controlled and regulated by the government, which subsequently vested the regulatory power in the ICAI through an enactment in the parliament in 1949. The enactment was, The Chartered Accountant Act of 1949, for the regulation of the profession of accounting in India. This act established a regulatory body for the Indian accounting profession in the form of the ICAI. The ICAI is responsible for conducting examinations, development of uniform standards of practice and the strategic development of the profession. It is also responsible for licensing, certification, training, and discipline (CAAC, 2002).

The ICAI has its headquarters at New Delhi with five regional offices at Mumbai, Chennai, Kanpur, Kolkata, New Delhi and 98 branches spread all over the country. In addition, it has also set up 11 chapters outside India. The total membership of the ICAI is about 115,000 and over 250,000 students are pursuing the Chartered Accountancy course. Out of the total members,



nearly 70% are in practice (www.icai.org). The Indian accounting profession is mainly comprised of small and medium sized firms, the number of firms with five or more partners being only about 375 in 2001 out of a total of 42,339 firms. The table below gives the distribution of firms by size in 2001 (CAAC, 2002).

Table 1. Size distribution of CA firms as at 01/04/2001

No of Partners	No of Firms
2	7161
3	2104
4	796
5	375
6	305
7	206
8	101
9	61
10	34
>10	52

The Big Four accounting firms entered the Indian market in the backdrop of the reform measures initiated by the Indian government due to the foreign exchange crisis in 1991. However, they could not provide statutory audit services due to provisions in the Indian Companies Act of 1956, which only allowed local accounting firms to conduct statutory audits. Therefore, the Big Four accounting firms entered into an arrangement with a few large Indian accounting firms, turning them into "surrogate firms" (CAAC, 2002:34). Each of the Big Four accounting firms



has more than one surrogate firm. Most of the surrogate firms are well established large Indian accounting firms in the traditional areas of audit and assurance services (www. icai.org). The list of the surrogate accounting firms of the Big Four accounting firms and dates of mergers are provided in the table below.³

Table 2. List of Surrogate Accounting firms

Multinational Accounting Firm	Year of Merger	Surrogate Indian Accounting
		Firm
KPMG	1995	Bharat S Raut & Co
Ernst & Young	1995	S R Batliboi & Co
Deloitte & Touche	1995	P C Hansotia & Co; Fraser &
		Ross; C C Choksi & Co
	1999	S B Billimoria
	2004	A F Ferguson & Co
Pricewaterhouse Coopers	1990	Pricewaterhouse; Lovelock &
		Lewis

Entry Pricing Literature

Limit pricing models emphasize deterring entry before it occurs: an incumbent firm setting its pre-entry price low enough to make entry appear unprofitable. Implicit in the limit pricing models is the belief that potential entrants use current industry profits as an indicator of future profits (Masson and Shanan, 1982). Such a strategy makes sense for incumbents if the pre-entry profits foregone are less than the additional profits earned by deterring entry and

³ For the purpose of this study, the surrogate firms will be treated as Big Four firms from the year of their merger with the Big Four accounting firms.

subsequently raising prices back to monopoly levels (Geroski, 1995). Early theoretical research, notably by Bain (1956), Modigliani (1958), and Sylos-Labini (1962) advocated limit pricing as an effective strategy for deterring entry. However, empirical evidence provides little support for limit pricing strategies. Smiley (1988) conducted a survey of U.S. firms and Singh et al. (1997) conducted a survey of U.K. firms. Both studies found that firms rarely use prices to deter entry. Dynamic limit pricing models, which emphasize that incumbents use their prices to monitor entry behavior continuously, also find little support empirically (Simon, 2005).

Game-theoretic models suggest that rational potential entrants will not be affected by low pre-entry prices, and thus incumbents only squander profits by limit pricing (Milgrom and Roberts, 1982a). These models emphasize the need for incumbents to credibly commit to low post-entry prices in order to deter entry. Another game-theoretic approach uses asymmetric information to make low prices a credible threat. Kreps and Wilson (1982) and Milgrom and Roberts (1982b) found that when the entrant is not sure about the incumbent's pay-offs, the incumbent might have an incentive to cut prices after entry as a way to build a reputation for fighting entry.

Several papers have studied incumbent pricing responses to entry, yielding inconsistent results. In the airline industry, Joskow, Werden, and Johnson (1994) find that incumbents cut prices following entry, while Windle and Dresner (1995) report that entry by low-cost carriers induces price cuts from incumbents. However, Thomas (1999) finds that incumbents in the ready-to-eat breakfast cereal industry do not cut prices after entry. Frank and Salkever (1997) find that brand-name prescription drug prices increase after generic entry, but generic prices fall with generic entry. Yamawacki (2002) finds that some car manufacturers reduce prices while some do not in response to entry. He argues that incumbent pricing responses to entry are firm-



specific, depending on the incumbent's ability to respond to entry, which in turn depends on its relative resource position.

In industries with differentiated products, firms have a variety of tools available to them for responding to new entry. These tools include advertising and promotional campaigns, and new product innovations, which may be used instead of or in addition to price responses (Thomas, 1999). There is also evidence that firms in highly differentiated industries use advertising to deter or limit entry (Thomas, 1999). But this argument fails to explain why incumbents in the airline and auto industry cut prices after entry (Windle and Dresner, 1995). Simon (2005) offers a more general explanation for why some firms respond to entry more aggressively than others: incumbents vary in their incentive to respond to entry. Several factors influence the incumbents' response to entry such as the incumbent's age, the incumbent's scope of services, and the market structure. But low entry barriers in the magazine market may explain the absence of a pricing response across all the incumbents in his study. Table 3 summarizes the empirical studies of incumbent pricing responses to entry.



Table 3. Empirical findings of incumbent pricing responses to entry

Authors	Industry	Key Result
Joskow, Werden & Johnson (1994)	Airlines	Incumbents cut prices post entry
Windle & Dresner (1995)	Airlines	Incumbent cut prices following entry
Marion (1998)	Grocery Stores	Incumbent supermarkets cut prices following entry by warehouse stores
Thomas (1999)	Cereal	Incumbents accommodate on prices
Frank & Salkever (1997)	Pharmaceuticals	Incumbent producers of prescription drugs raise prices after generic entry. Generic incumbents reduce prices
Bresnahan and Reiss (1991)	Local Retail Markets	reduce prices after entry. Incumbents reduce prices following entry
Yamawacki(2002)	Automobiles	Firm specific and Group specific factors influence response to entry
Simon (2005)	Magazines	Firm specific and Industry specific factors influence response to entry

Incumbent pricing expectation in an audit setting

Marketers have suggested that, while the basic concepts of marketing remain the same across industries, professional services firms have characteristics that set them apart from their



manufacturing counterpart (Crittenden et al, 2003). In the market for services, Monroe (1989) suggested that most organizations have traditionally followed a naive and unsophisticated approach to pricing. Murdock and Mcgrail(1994) go so far as to suggest that accountants 'avoid overtly using price as part of an overall marketing program'(pp.17). However, DeAngelo (1981) argued that prices are used by accounting firms to gain new clients and that rational auditors charge lower audit fees in the first few years of their engagement as they know that they will be able to earn excess profits in the later years, a strategy known as "lowballing". The longer the association of the auditor with the client, the more knowledgeable the auditor is about the operations of the client and its industry. Such knowledge is very helpful as it allows the auditor to improve audit efficiency and effectiveness. When bidding on a new engagement, rational auditors are aware that they have an opportunity to earn excess profits in the later years of their relationship. Therefore, auditors tend to reduce the fee bid in the early years to attract the clients. Ettredge and Greenberg (1990) actually found that the initial fees charged by the new auditors were about 25% lower than the last fees charged by their predecessors. Further, in the Indian setting, it can be argued that the Big Four accounting firms might adopt the strategy of "lowballing" to gain access to a client for the purpose of providing non-audit services or to gain recognition in the local market.

Along with the "lowballing" theory, there is also evidence that the Big Four accounting firms command price premiums (Francis and Simon, 1987; Palmrose, 1986). This premium has been primarily explained with two different arguments. First, DeAngelo argued that clients use the size of the accounting firm as a quality surrogate and therefore large accounting firms such as the Big Four accounting firms are perceived to supply higher levels of audit quality as compared to the smaller firms. She argues that audit quality is difficult to evaluate because of the inherent



nature of auditing. Thus, rational consumers devise alternative arrangements to find out the quality of audit services provided by different accounting firms and therefore they use auditor size as a surrogate for audit quality (DeAngelo, 1981). The other argument is based on the brand name model of Klein et al. (1981) in which the Big Four accounting firms are expected to develop and maintain brand name reputations for quality in order to secure and protect quasirents arising from the brand name (Johnson and Lys, 1990). In other words, the brand-name development comes first and this in turn leads to a quality-assuring price that is higher than the minimum-quality price.

Local accounting firms would be faced with either a "lowballing" strategy likely to be adopted by the Big Four accounting firms entering the market or a higher Big Four premium due to the perceived higher quality of services provided by the Big Four accounting firms and the brand reputation of the Big Four accounting firms. It is necessary to mention here that Pearson and Trompeter (1994) argued that the brand reputation of the Big Four accounting firms would reduce the likelihood of a "lowballing" strategy by the Big Four accounting firms. If the Big Four accounting firms adopt a "lowballing" strategy, the local accounting firms are more likely to lower their fees to meet the challenge posed by the Big Four accounting firms. However, when there are client specific start-up costs associated with the audit market, incumbent auditors have certain cost advantages over the potential competitors for a certain client. They are already well acquainted with the operations and accounting practices of the client and have qualified staff that has developed a working relationship with the management of the client firm, that is "incumbent auditors earn client specific quasi rents" (DeAngelo, 1981, pp.184). This suggests that the termination of the audit relationship would impose costs on both parties, the auditor and the client. In the event of termination, the incumbent auditors would lose the "wealth equivalent of



the client-specific quasi-rent stream" (pp. 188) and the clients would have to bear the transactions costs of seeking services of new auditors and start-up costs of training the new auditors to get them familiarized with the operations of the organization (DeAngelo, 1981). Further, there is a value placed by the clients on the higher quality of service and brand reputation of the Big Four accounting firms; say, the premium value. Therefore, I argue that the local accounting firms would lower their fees if they believe that the sum of the premium value and the discount offered by the Big Four accounting firms adopting a "lowballing" strategy would exceed the high direct costs likely to be incurred by the clients to switch auditors. On the other hand if rational auditors of the local accounting firms reckon that the transaction costs of switching and duplication of start-up costs associated with training a new auditor are higher than the sum of the premium value and the discount offered by the Big Four accounting firms, they would meet the challenge posed by the competitors by increasing the scale of their audit services or other non-pricing measures.

On the other hand, if the Big Four accounting firms charge a higher premium right from the outset due to their perceived higher quality of services and brand reputation, it would not make sense for the incumbent accounting firms to reduce their fees as their existing fees are expected be lower than the audit fees charged by the Big Four accounting firms. This course of action of the incumbent accounting firms would depend on the premium value. If the premium value is more than the sum of the fee difference between the Big Four accounting firm and the incumbent accounting firm and the transaction costs of switching auditors, the client would switch from the local accounting firm to the Big Four accounting firm. In such a scenario, the incumbent accounting firm could be expected to reduce its audit fees to retain the client as long as it is still making economic profits. This in turn would depend on the profit margins of the



incumbent accounting firms. Given the absence of foreign competition in the local accounting services market before the entry of the Big Four accounting firms, the margins of the incumbent accounting firms can be expected to be high enough to sustain a reduction in audit fees to deter entry.

There is little empirical evidence which provides support for limit pricing strategies and most of the empirical studies found that firms rarely use pre-entry pricing to deter entry. In case of pricing strategies predicted by the game theoretic models, low prices have to be maintained post-entry and when the entrant is not aware of the incumbent's pay-offs, the incumbent has incentive to cut prices post-entry to deter the entrant(Kreps and Wilson, 1982). This is particularly relevant in the Indian audit market, where the Big Four accounting firms are not likely to possess knowledge about the cost structure and profit margins of the local accounting firms. This would provide additional incentive to the local accounting firms to cut prices after entry to drive out the Big Four accounting firms.

Hypotheses and Research Methodology

Preliminary Analysis: Estimation of Big Four Premium

Before developing the hypotheses for the research questions in the study, it is important to analyze the relevance of the audit fee model in the Indian audit market because I intend to use the model to test the pricing response of the local accounting firms to the entry of the Big Four accounting firms. The market for audit services has been studied extensively in the developed countries including United States of America (Simunic, 1980), United Kingdom (Taylor and Baker, 1981), Australia (Simon and Francis, 1988), and in New Zealand (Firth, 1985). However, little work has been done in developing countries on the determinants of audit fees except for Simon et al. (1995) and Karim et al. (1996). Karim et al. (1996) analyzed the determinants of



audit fees in Bangladesh, whereas Simon et al. (1995) examined the determinants of audit fees for public sector companies in India and compared their fees with fees for private sector companies in the same market. Both the studies found the audit fee model to be relevant in India and Bangladesh.

In this study, for completion purposes, I intend to start with an analysis of the relevance of the audit fee model in the Indian audit market. This will allow for a direct comparison with the studies done in Hong Kong and the larger countries of America and Australia. If the audit fee variables which have been known to be determinants of audit fees in prior studies are also found to be related to audit fees in the Indian audit market, we can conclude that the audit fee model is relevant in the Indian audit market. For instance, total assets have been empirically found to be one of the determinants of audit fees and they have also been found to be positively related to audit fees. If total assets are, therefore, found to be positively related to audit fees in the Indian audit market, it can be concluded that the audit fee model is relevant in the Indian audit market.

Hypothesis 1: Price cutting by the local accounting firms

There are two kinds of pricing strategies adopted by incumbents in the face of entry: those that predict pre-entry pricing response and those that predict a post-entry pricing response. In the first category are limit pricing models in which incumbents set a pre-entry price below the current profit maximizing level as a way to reduce the potential entrants' assessment of the benefits of entry. In the second category are game-theoretic models in which incumbents reduce prices post-entry to drive out entrants and deter future entrants (Simon, 2005). In the Indian audit market, the Big Four accounting firms would either lower their initial fees to get new clients and gain entry, a strategy known as "lowballing", or due to their perceived higher quality of services and brand reputation, they would charge a premium for their services. If the Big Four accounting



firms adopt the latter strategy, the local accounting firms would be more likely to maintain their audit fees as their fees would already be at a discount compared to the higher fees to be charged by the Big Four accounting firms, except for the case when the premium value is higher than the sum of transaction costs of switching auditors and the fee difference between the Big Four accounting firm and the incumbent firm. In the latter case, the incumbent firms are more likely to reduce their audit fees to retain their clients. If the Big four accounting firms adopt a "lowballing" strategy, the local accounting firms would compare the transaction costs associated with switching auditors with the sum of the premium value and the discount offered by the Big Four accounting firms. If the transaction costs are higher than the sum of the premium value and the discount, the local accounting firms would not reduce their fees and if the transaction costs are lower than the sum of the premium value and the discount, the local accounting firms would lower their audit fees to deter the Big Four accounting firms.

In case of pre-entry pricing strategy, if economies of scale require entrants to achieve at least a minimum market share, and if they believe that incumbents will not change their pre-entry prices post entry, then an appropriate choice of pre-entry prices by the incumbent will ensure non-positive post- entry profits for the entrant. The limit pricing strategies make sense for incumbents if the foregone pre-entry profits are less than the present discounted value of the entire stream of post-entry profits, relative to what incumbents would have earned if entry had occurred (Geroski, 1995). This calculation, in turn depends on the speed with which entrants penetrate the market, on market growth, and above all, on the height of entry barriers. If the barriers to entry are high, firms will not have to cut prices much below monopoly levels to deter entry, while if barriers to entry are low then the market is effectively competitive and the best



strategy for incumbents would be to not cut prices and maximize the monopoly profits in the short run (Geroski, 1995).

In the Indian audit market, post the liberalization reforms in the early 1990s, the barriers to entry were considerably reduced in the consulting services of the market, although the audit market was still protected by legislation (CAAC, 2002). The local accounting firms would therefore be expected to reduce fees for their audit services, but not cut prices for consulting services, essentially non-audit fees. Further, the Big Four accounting firms would be expected to penetrate the market quite rapidly due to their brand reputation and perceived higher quality of service and therefore, the local accounting firms would be expected to forego some of their preentry profits to deter entry by the Big Four accounting firms. Another drawback of the limit pricing strategy is that while it makes sense to threaten to reduce prices pre-entry, an incumbent faced with the actual fact of entry will wish to maintain high prices(jointly with the entrant) postentry. In other words, the threat to maintain high pre-entry prices indefinitely in the post-entry future may not be persuasive to a sophisticated entrant, in this case, the Big Four accounting firms with a long history of entry in world markets.

In case of differentiated product markets, the incentive to cut prices is weaker because demand is more inelastic with respect to price and more elastic with respect to other marketing tools. In such cases, firms may not react with prices but respond aggressively with another competitive strategy (Gruca, Kumar, and Sudarshan, 1992). The intangible nature of accounting and auditing work results in some level of product differentiation in the audit market (Parasuraman et al., 1985; Schonberger, 1980). That is, Big Four accounting firms could be providing a different level of 'audit risk', 'assurance level', or 'insurance' as compared to a Non-Big Four accounting firm. However, firms in differentiated markets may also respond with more



than one competitive weapon (Gatignon and Hanssens, 1987). Extant marketing literature argues that incumbents in differentiated markets almost always reduce prices post-entry (Hauser and Shugan, 1983; Kumar and Sudarshan, 1988; Gruca et al., 1992). In light of the discussion above, I offer the following hypothesis in the alternative form:

Hypothesis 1: Local accounting firms significantly reduced their audit fees post-entry by the Big Four accounting firms.

Hypothesis 2: Effect of Market structure on the pricing response of local accounting firms

Market structure may also influence the incumbents' incentive to respond to entry. Prior research has shown that the threat posed by entry is greater in concentrated markets (Hannan, 1979). In competitive markets, entry should have little effect on incumbents as competition has already forced high prices down towards marginal cost, while in more concentrated markets, entry threatens to erode rents by making it more difficult to maintain tacit collusion. Therefore, incumbents in highly concentrated markets have a greater incentive to cut prices, both to drive out entrants and to defer further entry (Hannan, 1979; Kessides, 1990). Alternatively, it has also been argued that higher concentration may reflect entry barriers or incumbent capabilities which make it more difficult for new firms to enter the market (Simon, 2005). Hannan (1979) finds that incumbents in concentrated markets cut prices more in response to entry. In 2001, the Indian audit market was comprised of 52 medium sized accounting firms (No of partners > 10) and approximately 42,000 small sized accounting firms (CAAC, 2002). Market competition induces clients and accounting firms to align themselves to achieve efficient utilization of specialized resources and therefore, accounting firms that normally audit small, unregulated clients may be unable to offer their services at competitive prices to large, geographically dispersed corporations because they lack the economies available to firms already serving in the



market segment (Johnson and Lys, 1990). Conversely, firms that typically audit large geographically dispersed clients will be unwilling to allocate productive resources to small localized corporations at competitive prices. Auditors of comparable size and clientele mix can thus be expected to have similar cost structures (Johnson and Lys, 1990). It follows that medium sized accounting firms are more likely to compete for large Indian companies for audits and therefore, the market for large auditees is likely to be highly concentrated due to the small number of medium sized accounting firms. Conversely, the market for smaller auditees is likely to be more competitive due to the large number of small sized accounting firms operating in the Indian market. It is reasonable to expect the Big Four accounting firms to target the largest companies in the Indian market to penetrate the market. It follows that the medium sized firms are more likely to be threatened by the entry of the Big Four accounting firms and therefore are more likely to reduce prices, whereas the small sized firms are not likely to reduce prices as competition has already forced down prices in the small auditee segment of the Indian market.

Hypothesis 2: Local medium sized accounting firms y reduced audit fees significantly more than local small sized accounting firms in response to entry.

Sample Selection and Model Specification

To test these hypotheses, I use data from the Indian audit market. Per the provisions of the Indian Companies Act 1956, all public companies are required to disclose audit fees and non-audit fees in annual reports. The data span a ten year period from 1996 to 2005. The sample includes 206 randomly selected publicly held companies from the Bombay Stock Exchange -500 Index (hereafter, BSE-500). In total, there are 2060 client-year observations. Finance companies and Banks will be excluded from the sample selection process, as it has been established before



that most of the financial ratios used in the model cannot be determined in case of finance companies (Simunic, 1980).

Regression analysis will be the major statistical technique utilized in the empirical analysis. Two OLS regression models are to be specified:

(1) First, to test the relevance of the audit fee model in the Indian audit market, a cross-sectional audit fee model will be used for the full sample of companies for the year 2005. These audit fee models use a variety of variables to control for different factors that have an impact on audit fees such as client size, audit complexity, and auditor-clients risk sharing (Simunic, 1980). Simon and Francis (1988) and Chan et al. (1993) have shown that these models have demonstrated relatively high explanatory power (R² of 0.65 or higher) and have been robust across different samples, time periods, and countries, and to sensitivity analyses for model misspecification.

Model 1:

$$LAUD = b_1 + b_2LTOT + b_3LOSS + b_4DE + b_5CARAT + b_6IAUDITOR1 + b_7$$

$$IAUDITOR2 + b_8SUBS$$

Where,

LAUD = the natural logarithm of audit fees

LTOT = the natural logarithm of total assets

CARAT= current assets divided by total assets

DE = long-term debt divided by total assets

LOSS = indicator variable (1= loss reported in current or prior year)

IAUDITOR1 = indicator variable (1= Medium local accounting firm)



IAUDITOR2 = indicator variable (1= Small local accounting firm)

SUBS = Square root of subsidiaries

(2) Second, to investigate the pricing response of incumbent firms, a longitudinal fixed effects audit fee model will be used for the full sample for the entire data span (1996 to 2005). In the fixed effects specification, only within- accounting firm variation and within-client variation is used. In this model, the purpose is to compare the audit fees charged by incumbent accounting firm to its client at the time of entry, which is the year, 1996 with the audit fees charged by the same accounting firm after entry over the period of 10 years, averaging the difference across all the incumbent accounting firms in the sample.

Model 2:

 $LAUD = b_1 + b_2LTOT + b_3LOSS + b_4DE + b5CARAT + b_6ROI + b_7IAUDITOR1 + b_8IAUDITOR2 + b_9ENTRY$

Where,

LAUD = the natural logarithm of audit fees

LTOT = the natural logarithm of total assets

CARAT = current assets divided by total assets

ROI = Net Income divided by total assets

DE = long-term debt divided by total assets

LOSS = indicator variable (1= loss reported in current or prior year)



IAUDITOR1⁴ = indicator variable (1= Medium local accounting firm)

IAUDITOR2 = indicator variable (1= Small local accounting firm)

ENTRY= indicator variable (1 = 1996 to 2000)

⁴ For this variable, the local accounting firms which merged with the Big Four accounting firms will be classified as incumbent accounting firms till the year of merger and after year of merger, they will be classified as Big Four accounting firms.



Table 4. Models and Hypotheses

Hypotheses	Models	Variables of Interest	Expected sign
Preliminary	Model 1	LTOT	+
		CARAT	+
		DE	+
		LOSS	_
		SUBS	+
		IAUDITOR1	-
		IAUDITOR2	-
Hypothesis 1	Model 2	ENTRY	_
Hypothesis 2	Model 2	IAUDITOR1 > IAUDITOR2	-

Table 4 provides a description of all the hypotheses and models used in the study along with the expected signs of the variables of interest. In Model 1, LTOT, CARAT, DE, LOSS, IAUDITOR1 and IAUDITOR2 are the variables of primary interest and are intended to assess the relevance of the audit fee model in the Indian audit market. Based on previous studies, LTA is intended to control for client size, while CARAT, SUBS and DE control for audit complexity and risk. All four variables are expected to be positively related to audit fees. LOSS is expected to be negatively related to audit fees. IAUDITOR1 and IAUDITOR2 are intended to assess the extent, if any, of a Big Four premium in the Indian audit market (Davis et al., 1993; Firth, 1985; Simunic, 1980). IAUDITOR1 has been defined as any local accounting firm with more than fifteen partners, whereas IAUDITOR2 has been defined as any local accounting firm with less than ten partners⁵. Logarithmic transformations of fees and assets are employed because previous

المنسارة للاستشارات

⁵ None of the local accounting firms in the sample had between 10 and 15 partners and therefore I classified the small and medium sized accounting firms based on the above limits.

research in this area (Simon and Francis, 1988) has indicated that this specification provides a good linear fit in which the assumptions of ordinary least squares regression are satisfied.

In Model 2, I measure entry with a dummy variable (1 = 1996 to 2000) for all the incumbent firms in the sample and this variable is expected to be negatively related to LAUD, indicating that the incumbent auditors reduced prices following entry by the Big Four accounting firms (Hypothesis 1). Further, the coefficient size of IAUDITOR1 will be compared with the coefficient size of IAUDITOR2 to determine whether market structure moderates the pricing response of the incumbent firm (Hypothesis 2).

Results

Preliminary Analysis: Estimation of Big Four premium

Table 5 presents the results of the preliminary analysis conducted to examine the effectiveness of the audit fee model in the Indian market. The cross sectional sample includes all the 206 firms from the BSE-500 index for the year 2005. The variables used to estimate audit fees are the same variables used in prior audit market studies (Simunic, 1980; Craswell et al., 1995). Prior studies (mainly using U.S. and Australian data) have documented a brand-name premium for the Big 6 auditors. Simon and Francis (1988) report a premium of 16 percent of total audit fees in a study using U.S data and calculate that the Big 6 premium averages 18% across a number of other studies using either U.S or Australian data. Craswell et al. (1995) use a much larger sample and estimate a Big 6 premium of 31% in Australia. We replicate and extend the design used by Craswell et al. (1995). The model is significant at P< 0.01 and has an adjusted R² of 39%. The IAUDITOR1 variable is negative but statistically insignificant, whereas the IAUDITOR2 variable is negative and statistically significant, which means that a premium is



paid to the Big Four auditors in India as compared to the small sized accounting firms. The coefficient of -0.336 in the regression estimation translates to a premium of 48%.⁶

Table 5. Preliminary Analysis: Estimation of Big Four premium

Variables	Coefficient	t-statistic	Probability
Intercept	2.044	4.305	0.000
LOGTOT	0.553	8.868	0.000
LOSS	0.148	2.420	0.016
DE	-0.145	-7.81	0.000
CARAT	0.135	5.870	0.000
IAUDITOR1	-0.092	-1.504	0.134
IAUDITOR2	336	-5.484	0.000
SUBS	0.185	3.341	0.001
F-Statistic	19.129	0.000	
Adjusted R ²		39%	

This is higher than the average 16 to 18 percent premium in U.S. studies (see Simon and Francis 1988) and also higher than the Big 6 premium of 31% in Australia (Craswell et al., 1995). This can be explained by the absence of many larger sized Non-Big Four accounting firms in the

⁶ A procedure to derive the impact of the auditor intercept shift on the dependent variable (which is the natural log of audit fees) is reported by Simon and Francis (1988, 263).

Indian market. As noted elsewhere in the paper, the number of firms with partners greater than 10 is only about 42 out of the total number of around 42,000 firms. Defond et al (2000) also found that the Big 6 premium in Hong Kong was much larger than the premium found in U.S. and Australian studies due to the same reason.

Incumbent Pricing Results

Table 6 provides descriptive statistics and a correlation matrix. Looking at the sample means, the incumbent pricing responses are being measured for the first six years of the time period from 1996 to 2005, about 44%. The medium sized accounting firms represent about 23% of the sample while the small sized accounting firms represent about 46% of the sample, the rest being the Big Four accounting firms. The correlation matrix in Table 6 is a 'within' correlation matrix. That is, the correlations reflect only the variation within each panel (client). This corresponds to the fixed-effects model which only exploits the variation within each panel (client). As can be seen from the correlation matrix, multicollinearity does not seem to be a problem⁷.

⁷ All the VIF reported were less than 1 in the entire sample.



Table 6. Descriptive statistics and within-client correlation matrix

Variable	Mean	S.D	1	2	3	4	5	6	7	8	9	10
LOGAUD	5.76	.51	1.00	.53	.14	.04	15	.003	33	.16	29	.24
LOGTOT	9.67	.62	.532	1.0	126	.046	.061	386	073	.111	.014	.068
ROI	.10	.12	.146	126	1.000	271	339	.285	136	001	110	.012
LOSS	.06	.24	.041	.046	271	1.000	.170	068	033	.030	024	.015
DE	.38	.31	150	.061	339	.170	1.00	193	.080	.016	.101	009
CARAT	.69	.41	.003	386	.285	068	193	1.000	109	067	116	.012
ENTRY	.44	.49	330	073	136	033	.080	109	1.000	.220	.294	.006
IAUDITOR1	.23	.41	.166	.111	001	.030	.016	067	.220	1.000	505	.090
IAUDITOR2	.46	.49	292	.014	110	024	.101	116	.294	505	1.000	085
SUBS	1.96	3.579	.240	.068	.012	.015	009	.012	.006	.090	085	1.000

Table 7 reports results for the model used to test the hypotheses 1 & 2. In this model, the coefficient for ENTRY is negative and statistically insignificant providing support for Hypothesis1 and suggesting that, on average, incumbent accounting firms cut prices post entry by the Big Four accounting firms in the Indian audit market. The coefficient value is -0.19, which suggests that local Indian accounting firms, on average, cut prices by around 19% in the first five years from 1996 to 2000. The variables of IAUDITOR1 and IAUDITOR2 are also negative and statistically significant with values of -.12 and -.40 respectively, thereby failing to provide support for hypothesis 2. However, the result suggests that small sized local accounting firms cut prices more than the medium sized accounting firms. While the small sized accounting firms reduced prices by around 40% over the ten year period, the medium sized accounting firms reduced prices only by around 12% over the same period. The reasons for the same will be discussed in the next section.



Table 7. Incumbent Pricing- Audit fee estimation from 1996-2005

Variables	Coefficient	t-statistic	Probability	
Intercept	-1.226	-5.560	0.000	
LOGTOT	0.755	33.910	0.000	
LOSS	0.044	2.090	0.030	
DE	-0.145	-7.810	0.000	
CARAT	0.135	5.870	0.000	
ROI	0.212	0.048	0.000	
ENTRY	-0.194	-17.39	0.000	
IAUDITOR1	-0.124	-4.330	0.000	
IAUDITOR2	-0.402	-8.380	0.000	
SUBS	0.008	0.790	0.428	
F-Statistic	29.950	0.000		
Adjusted R ²	88%			



Discussions and Conclusions

Economic theory suggests that incumbent firms may cut prices after entry either to deter future entrants or to maximize current profits. However, empirical research so far has produced inconsistent results: some studies, including the current paper, find a negative relationship between entry and pricing, while others do not.

Regarding the direct effect of entry, the results of this study indicate that entry does have a statistically significant effect on incumbents' pricing in the Indian audit market. One industry-specific explanation for the effect of entry may be the relatively low rate of entry in the audit market due to entry barriers. Because entry was rare in the Indian audit market, the incumbent accounting firms were motivated to cut prices to protect their monopolistic profits (Hannan, 1979; Cool, Roller, and Leleux, 1999). In other words, prior to the entry of the Big Four accounting firms, the Indian audit market was a protected market in which incumbents were charging high audit fees not fearing entry in the market. Another explanation for the significant pricing response could be the emphasis on the long term effects of entry, around 10 years, in this paper as opposed to other studies which mainly examined the short run effects of entry (Geroski, 1995).

Hypothesis 2 was not supported. In fact, the results indicate that small sized accounting firms cut prices more than the medium sized accounting firms. There can be various reasons for these results. Firstly, as discussed in the earlier part of the paper, the pricing responses of the local accounting firms would depend on a large extent on the premium value assigned to the Big Four accounting firms by the clients of the local accounting firms. It can be reasonably asserted that the premium value assigned to the Big Four accounting firms by clients of the small sized accounting firms would be higher than the premium value assigned by the clients of the medium



sized accounting firms. Therefore, it follows that the small sized accounting firms might have been more threatened by the entry of the Big Four accounting firms than the medium sized accounting firms and might have been more inclined to cut prices to deter entry. Further, the sample of firms used in the paper could also provide some explanation for the results of Hypothesis 2. The sample of firms was randomly selected from the BSE- 500 index, which would consist of the largest firms in the Indian market. These firms can be expected to contribute a significant proportion of the audit fee revenue for the small sized accounting firms. It follows therefore that the small sized accounting firms would be motivated to retain these clients and therefore would be more likely to cut prices for these clients.

Another explanation for the more aggressive pricing responses of the small sized accounting firms as opposed to the medium sized accounting firms could be explained by the argument by Smith, Grimm and Gannon (1992), who suggested that firms with more complex organizational structures are less likely to respond to competitive attacks and entry. They posit that in structurally complex firms, decision makers receive information more slowly and are more likely to receive misinformation. This weakens the ability of these firms to respond to entry. The lack of aggressive pricing responses of the medium sized accounting firms may be explained by these organizational factors.

Finally, while Hypothesis 2 was based on the argument that incumbents in concentrated markets cut prices more in response to entry, it has also been argued that higher concentration may reflect entry barriers or incumbent capabilities which make it difficult for new firms to enter the market. In this case the entry barriers would grant the incumbents a competitive advantage over new entrants, reducing the need for incumbents to respond aggressively to entry (Simon, 2005).



There are some limitations in this paper which need to be addressed. One concern with this study is that local accounting firms may respond to entry with non-price weapons. For instance, incumbent firms may increase their advertising expenditures or upgrade the quality of their services or provide more services at the same price. These non-price responses however, may tend to be cost increasing, leading to price increases. As a result, these non-price responses may cause a spurious, positive relationship between entry and price (Simon, 2005). This may create a bias against finding a negative relationship between entry and price, thereby strengthening the results found in this paper.

Another drawback in this paper is the absence of non listed companies in the sample. It could be possible that the small sized accounting firms reduced prices for their listed clients but not for the small privately held companies. It would be interesting to examine the responses of the local accounting firms in the private sector of the Indian industry. One might also be concerned that changes in supply conditions, for example, a reduction in costs due to technological innovations could lead to lower prices, yielding a downward bias in the coefficient on entry (Geroski, 1995). However, this is unlikely in a single industry setting because factors that reduce production costs would be likely to affect all the accounting firms, regardless of the type of client they audit. This effect would be captured by the year fixed effects (Simon, 2005).



References

- Bain J., 1956, Barriers to New Competition, *Harvard University Press*, Cambridge, MA.
- Chan, P., M. Ezzamel, and D. Gwilliam, *Journal of Business Finance & Accounting*, 20, 765-787.
- Crittenden, V., R. Davis, D.T.Simon, and G. Trompeter, 2003, *Journal of Strategic Marketing*, 37-53.
- Cubbin, J., and S. Domberger, 1988, Advertising and post-entry oligopoly behavior, *Journal of Industrial Economics*, 123-140.
- The Chartered Accountants Action Committee For Level Playing Field, 2002, White paper on Multinational Accounting Firms operating in India, 1-141.
- DeAngelo, L. E., 1981, Auditor Size and audit quality, *Journal of Accounting and Economics*, 183-200.
- Dierickx, I., and K.Cool, 1989, Asset stock accumulation and sustainability of competitive advantage, *Management Science*, 1504-1511.
- Ettredge, M., and R. Greenberg, 1990, Determinants of fee cutting on initial audit engagements, *Journal of Accounting research*, 198-210.
- Firth, M. 1985. An Analysis of Audit fees and Their Determinants in New Zealand. *Auditing, A Journal of Practice and Theory*, 23-37.
- Fombrun, C., and M. Shanley, What's in a name? Reputation-building and corporate scope, *Academy of Management Journal*, 233-258.
- Francis, J., and D. Simon, 1987, A test of audit pricing in the small-client segment of the U.S. audit market, *The Accounting Review*, 145-157.
- Frank, R., and D. Salkever, 1997, Generic entry and the pricing of pharmaceuticals, *Journal of Economics and Management Strategy*, 75-90.
- Gatignon, H., and D. Hanssens, 1987, Modeling marketing interactions with application to salesforce effectiveness, *Journal of Marketing Research*, 247-257.
- Geroski, P., 1995, What do we know about entry, *International Journal of Industrial Organization*, 421-440.
- Gruca, T., R. Kumar, and D. Sudarshan, 1992, An equilibrium analysis of defensive response to entry using a coupled response function model, *Marketing Science*, 348-358.



- Hannan T., 1979, Limit pricing and the banking industry, *Journal of Money, Credit and Banking* 11, 438-446.
- Hauser, J., and M. Shugan, 1983, Defensive marketing strategies, *Marketing Science*, 319-360.
- Johnson, W. B., & L. Thomas, 1990, The market for audit services: Evidence from voluntary auditor changes, *Journal of Accounting and Economics*, Elsevier, vol. 12(1-3), pages 281-308, January.
- Joskow, A., G. Werden, and R. Johnson, 1994, Entry, exit, and performance in airline markets, *International Journal of Industrial Organization*, 457-471.
- Karim, A.K.M., and P. Moizer. 1996. Determinants of Audit fees in Bangladesh. *The International Journal of Accounting*, 497-509.
- Kessides I., 1990, Towards a testable model of entry: a study of the U.S. manufacturing industries, *Economica* 57, 219-238.
- Klein B., and K.B. Leffler, 1981, The role of market forces in assuring contractual performance, *Journal of Political Economy*, 615–641
- Kreps, D., and R. Wilson, 1982, Reputation and imperfect information, *Journal of Economic Theory*, 252-279.
- Kumar, R., and D. Sudarshan, 1988, Defensive marketing strategies: an equilibrium analysis based on decoupled response function models, *Management Sciences*, 405-415.
- Masson, R., and J. Shannan, 1982, Excess capacity and limit pricing: an empirical test, *Economica*, 365-378.
- Milgrom, P., and J. Roberts, 1982b, Predation, reputation, and entry deterrence, *Journal of Economic Theory*, 280-312.
- Milgrom, P., J. Roberts, 1982a, Limit pricing and entry under incomplete information: an equilibrium analysis. *Econometrica*, 443-459.
- Modigliani, F., 1958, New Developments on the oligopoly front, *Journal of Political Economy*, 215-232.
- Monroe, K., 1989, The pricing of service, In C.A Congram and M.L. Friedman (eds), *Handbook of Services Marketing*, 5-18.
- Murdock, G., and G. Mcgrail, 1994, Developing a customer-oriented approach to fee setting, *The National Public Accountant*, 17-23.



- Nonaka I. 1994, Dynamic theory of organizational knowledge creation, *Organization Science* 5, 14-37.
- Palmrose, Z., 1986, Audit fees and Auditor size: Further Evidence, *Journal of Accounting research*, 97-110.
- Parasuraman, A., V. Ziethaml, and L. Berry, 1985, A conceptual model of service quality and its implications for future research, 41-50
- Rathmell, J., 1974, Marketing in the Service Sector, Cambridge, MA, Winthrop.
- Schonberger, R., 1980, Purchasing intangibles, *Journal of Purchasing and Materials Management*, 25-27.
- Simon, D. T., R. Ramanan, and A. Dugar. 1995. A comparative study of the Audit services market for public sector versus private sector companies in India. *Advances in International Accounting*, 1-13
- Simon, D., 2005, Incumbent pricing responses to entry, *Strategic Management Journal*, 1129-1248
- Simon, D., and J. Francis, 1988. The effects of auditor change on audit fees: Tests of Pricecutting and price recovery, *The Accounting Review*, 255-269
- Simunic, D. A. 1980, The Pricing of Audit Services, Theory and Evidence, *Journal of Accounting Research*, 161-190.
- Smiley, R., 1988, Empirical Evidence on strategic entry deterrence, *International Journal of Industrial organization*, 227-251.
- Sylos-Labini, S., 1962, Social Structure and organization, *In Handbook of Organizations*, 142-193
- Taylor, M.E., and R. L. Baker. 1981, An Analysis of the External Audit fee, *Accounting and Business Research*, 55-60.
- The New Indian Express, 2002, Publication of the Express Group of India, 14-17
- Thomas, L., 1999, Incumbent firms' response to entry: price, advertising, and new product introductions, *International Journal of Industrial Organization*, 527-555.
- Windle, R., and M. Dresner, 1995, Competitive responses to low cost carrier entry, Transportation Research: Part E, *Logistics and Transportation Review*, 59-75.
- Yamawacki, H., 2002, Price reactions to new competition: a study of U.S. luxury car market, *Industrial Journal of Industrial Organization*, 19-39.



CHAPTER THREE: STUDY TWO: IMPACTS OF GLOBALIZATION: EVIDENCE FROM THE INDIAN ACCOUNTING PROFESSION

Introduction

In the last two decades, globalization has become an integral part of the development strategies of many countries in Eastern Europe and Asia. Even though the apparent goals of globalization are world trade and investment without any barriers, it has been documented that globalization has only been used by the stronger nations to subjugate the weaker nation states (Caramanis, 2002). There has been a lot of prior research on the role of international agencies in the regulation of accountancy (Chua and Poullaos, 1993; Puxty et al., 1987; Sikka and Wilmont, 1995). Prior research has highlighted the control exercised by the British and American accounting associations over the development of indigenous accounting professions in colonial and post-colonial societies (Anisette, 2000; Caramanis, 1999; Chua and Poullaos, 1998, 2002; Johnson, 1973). However, as Caramanis (2002) points out, none of the above mentioned research efforts have analyzed the impact of globalization on the local accounting practices, discourses and institutions, especially those that deal with contemporary processes and events (Hopwood, 1994; Moizer, 1992). Also, Arnold (2005) reiterates that further case study research is needed to show how the process of globalization will affect local accounting practices and institutions and how the World Trade Organization (hereafter, WTO) members from nations outside the metropolitan center such as India, Brazil, and China will adapt to "transatlantic hegemony" (2005:323).

This main thesis of this paper is to respond to the calls by Caramanis (2002) and Arnold (2005) to enforce the view that globalization is not a natural and inevitable process, but that it is



a politically constructed institution that is driven by powerful international bodies such as multinational accounting firms and the International Monetary Fund (hereafter, IMF), using the instruments of coercion and pressure. Specifically, I intend to explore the interconnectedness of professional and state politics with global forces by examining the role played by professional associations and the state in the process of globalization.

While Caramanis (2002) examined the effect of developing patterns of global interconnections upon accounting decision making at the level of the nation-state, I intend to expand the analysis to examine the impact of globalization on the regulation of accounting by professional associations. These associations act as negotiating or representative agencies, shaping and defining appropriate practices of interaction for their members (Greenwood et al., 2002). They also act as the means whereby communities represent themselves to people outside the field. This role of professional associations makes them active players in the process of globalization. These associations along with the local government and the regulatory agencies are one of the first frontiers encountered by the forces of globalization in the accounting field.

The empirical portion of the paper is undertaken in the context of the accounting profession in India from 1990 to 2005. During this period, the Indian accounting profession underwent major changes both in terms of its market structure and jurisdiction. The change was precipitated by the foreign exchange crisis that almost turned the country bankrupt in 1991. To deal with the shortage of foreign exchange reserves leading to non payment of external debt, the government of India requested a Stand-By Arrangement from the IMF in August 1991 (Clark et al., 2003). One of the stipulations of the Stand-By Arrangement was that the multinational accounting firms ¹, essentially the Big Four accounting firms, be allowed to provide consulting

¹ The terms "multinational accounting firms" and "Big Four accounting firms" refer to the same constituents and therefore are used interchangeably throughout the paper.

services in India (CAAC, 2002). Consequently, the Reserve Bank of India issued special licenses to the multinational accounting firms authorizing them to provide consulting services in India. The analysis will focus on the catalystic role played by powerful international agents over the jurisdiction of the Indian accounting profession and the ability of these actors not only to overcome the resistance of local professional bodies but also influence the policies of the local regulatory agencies, banks, and financial institutions to the prejudice of the local accounting profession in a weak nation state such as India. The results will show how the multinational accounting firms, "well positioned in the international economic system" (Caramanis, 2002:380) are capable of not only changing the boundaries of the local accounting profession, but also becoming competitive forces in the market in a relatively short span of time from 1990 to 2005.

The paper mainly draws upon an academic exercise by the Chartered Accountants Action Committee For Level Playing Field(hereafter, CAAC) about the entry of the multinational accounting firms in India in the form of a white paper, which was published in 2002. The CAAC emphasized that the white paper was published in the "public interest to inform the Indian industry, Indian finance sector, Indian Government, Indian policy makers, Indian professionals, and also the general public about the correct facts of the multinational accounting firms (MAFs) and about the state of the Indian profession." (CAAC, 2002: 2). The white paper points out that the change in attitude of the Indian government towards globalization and liberalization in the early 1990s was not a deliberated national decision, but it was compelled and driven by the foreign exchange crisis that almost turned the country bankrupt in 1991. Further it is argued that an impulsive reaction to this crisis resulted in a defocused liberalization program, without seeking the inputs and opinions of the stakeholders who would be affected by



the liberalization agenda, such as the Indian companies, the Institute of Chartered Accountants of India (hereafter, ICAI), or the Indian accounting firms (CAAC, 2002).

The remainder of the paper is organized into the following sections. Section II looks at the prior research in the field of globalization and summarizes the insights from a diverse body of academic literature which forms the basis for my analysis. Section III describes the research methodology and provides a brief background of the Indian accounting profession. Section IV presents an analysis of the empirics examining the entry of the multinational accounting firms in India, the role played by the ICAI in the process of entry, and highlights certain theoretical issues. Finally, Section V summarizes the analysis, draws together the main points of the discussion and provides directions for future research.

Theoretical Context

Theoretical perspectives on globalization

Giddens defined globalization as "a process of increasing interconnectedness between societies in a dialectical fashion such that events in one part of the world more and more have effects on peoples and societies far away and vice versa" (1990: 64). Research on globalization is voluminous and diverse in content as the concept has been studied from various perspectives. These different perspectives include economy and polity (Camillieri and Falk, 1992; Dicken, 1992; Held, 1991), international relation and politics (Cox, 1996), trading blocks and the global financial system (Frieden, 1989), sociology and culture (Featherstone, 1990), domestic regulation and the market for professional services (Caramanis, 2002; Arnold, 2005). Some extreme views signal the end of the nation-state (Ohmae, 1996; Camilleri & Falk, 1992), while some views emphasize the state's resistance to attacks on its sovereignty by international agents (Hirst & Thomson, 1992, 1995). The common thread among all the diverse topics is the



consensus that there has been a remarkable shift of power from the sovereign state to multinational economic institutions such as the IMF and the World Bank. (Chase-Dun, 1989; Dicken, 1992; Kennedy, 1993). The extent of sovereign power exercised by a state can be determined by understanding the gap between the political power proclaimed by them and the actual practices and the economic system at the national regional and global levels. (Caramanis 2002). Thus from a domestic perspective, one needs to look at the interests of the local political actors, and corporate bodies influencing the decisions of the nation state, whereas from an international perspective, the power of the nation state to exercise sovereignty needs to be studied in relation to the world economy and international economic and military alliances, which set constraints on the sovereign power (Caramanis 2002). The current paper plans to examine the event of the entry of the multinational accounting firms from an international perspective on globalization. In other words, the analysis will focus on the influence of international economic institutions such as the IMF and multinational bodies such as the multinational accounting firms on the sovereignty of a weak nation state such as India and consequently on the local accounting profession.

There are four themes in the study of globalization examined by Held (1991) and clarified by Caramanis (2002) which are relevant to the current paper. First, it is argued that multinational or transnational corporations have become 'floating' international agents. While it has been documented that transnational corporations have strong links with their country of origin (Cox, 1996), it is important to note that a multinational business is influential enough within a major nation-state to influence the policies of that state in an attempt to expand internationally and make an entry into markets of weaker nation-states (Caramanis, 2002). In turn, these major nation-states exert pressure on international organizations such as the IMF or



the World Bank and use international trade agreements to exert influence on the weaker nationstates such as India, which would wish to protect its own markets (Arnold, 2005). Second, it has
been argued that while some nation-states such as the USA have the power and the resources to
resist international pressures on their economic policies, weaker nation-states such as India in
this case are more vulnerable in the management of their national economy (Cooper, 1986;
Gilpin, 1987). Third, some neorealist views emphasize a benign outcome of domination of one
state by another: the achievement of international order (Keohane, 1980). In a sense, these
hegemonic states serve as the establishers and underwriters of public goods, such as order and
openness and the existence of these states is necessary for the development of a world market
economy (Gilpin, 1987). Finally, these hegemonic states have the power to support and maintain
the international regime they favor, using a variety of coercive tools such as enforcement,
enticements or bribes to convince other states to remain within the confines of the regime
(Caramanis, 2002).

Theoretical Model: Institutional model of globalization

In the backdrop of the 'globalization' economy described above, the current paper employs the institutional model of globalization (Arnold, 2005) to examine the entry of the Big Four accounting firms in India. The conventional wisdom, which Arnold (2005) calls the market model of globalization argues that advances in technology have precipitated the growth of global financial markets. These global markets, in turn, require transparency in financial reporting to function effectively. This goal is served by the accounting and auditing disciplines by mitigating agency problems and reducing transactions costs. The role of the accounting profession in this model is reactive: the accounting profession must adapt and modernize in order to facilitate the smooth functioning of markets. This model, therefore, argues for adaptive responses ranging



from global professional credentialing to adoption of international accounting standards. The market model considers resistance to change on the part of nations-states and the accounting profession in these nation- states as rigidity that must eventually give way to the inevitable forces of globalization.

The institutional model of globalization, on the other hand, argues that the multinational accounting firms, with the help of their strong nation states of origin and international economic institutions, have worked proactively to create a global market for accounting and auditing services. In this model, the accounting profession plays an active role in the creation of global markets. The process of globalization begins with the agency of states, transnational corporations, and industry lobbies and culminates in the creation of global markets. In this view, non-market actors have deregulatory agendas, which motivate them to construct global markets to promote the dismantling of regulatory barriers to trade and investment (Arnold, 2005). The analysis in this paper will show the institutional model of globalization was in action in the Indian subcontinent enabling the entry of the multinational accounting firms into India through their global financial nexus. Further, the evidence in this paper will also show how local accounting practices, which were not sufficiently organized to face the challenges of globalization, give in to the market strategies and the combined strength of the multinational accounting firms. The evidence will also highlight the failure of the local government to protect the interests of the local accounting profession and in fact, the collusion of the local government with the multinational accounting firms to the prejudice of the Indian accounting profession.



Methods

Site and Data Sources

The field of interest in this paper is the provision of accounting services in the Indian market. This field is occupied by several organizational communities, including local accounting firms, the Big Four accounting firms, clients, and regulators such as Securities and Exchange Board of India(hereafter, SEBI) and the ICAI. The provision of accounting services in India is regulated by the ICAI and therefore these regulations can be used to demarcate the organizational community. The data sources used are interviews and archival materials. The main archival source is an extensive report about the entry of the Big Four accounting firms in the Indian market and its effects on the Indian accounting profession, in the form of a white paper published by the Chartered Accountants Action Committee For Level Playing Field. The CAAC emphasized that the white paper was published in the "public interest to inform the Indian industry, Indian finance sector, Indian government, Indian policy makers, Indian professionals, and also the general public about the correct facts of the multinational accounting firms (MAFs) and about the state of the Indian profession" (CAAC, 2002: 2). The white paper describes in detail the chronology of events from 1990 onwards about the entry of the Big Four accounting firms in the Indian market. Table 8 below provides a timeline of the events leading to and subsequent to the entry of the multinational accounting firms in the Indian market.



Table 8. Timeline of events in the process of globalization of the Indian accounting profession.

Year	Event
1949	The Chartered Accountants Act, 1949 was passed on 1st May. The term
	Chartered Accountant came to be used in place of Indian Registered
	Accountants.
	Chartered Accountants Act was brought into effect on 1st July. The Institute
	of Chartered Accountants of India is born.
1990	Foreign Exchange crisis in India and consequent probability of default on its
	borrowings from international markets and banks.
1991	To deal with the foreign exchange crisis, Government of India enters into a
	Stand-By Arrangement with the IMF.
	Structural adjustment program featuring macroeconomic stabilization and
	structural reforms.
1995	Reserve Bank of India grants special licenses to multinational accounting
	firms under Section 29 of the Foreign Exchange Regulation Act, 1947 to
	provide consulting services in India
1996 &1997	Multinational Accounting firms enter into an arrangement with large local
	accounting firms in India creating surrogate firms to gain entry in the audit
	and assurance services market
2002	Birth of the Chartered Accountant Action Committee For Level Playing Field
	(CAAC).
	Publication of the White Paper by the CAAC to shed light on the consequence
	of the entry of the multinational accounting firms in the Indian market.
2006	Chartered Accountant (Amendment) Act is enacted by the Central
	Government of India in the Gazette of India dated 23 rd March, 2006
	Removal of restrictions on sharing of profits by between members of the ICAI
	and non-members.
	Permission granted to provide multi-disciplinary services by accounting firms
	is granted
	Prohibition of solicitation is removed
	Limited advertisement (subject to guidelines to be issued by the ICAI) is also
	now permitted

The paper follows a research strategy which involves a mix of data and across-method triangulation (Campbell and Fiske, 1959; Denzin 1978; Jick, 1979). The paper relied extensively on the materials contained in the white paper and other publications of the CAAC, as well as on interviews. The interviews were semi-structured and aimed at obtaining information about the



key dimensions of the intra-professional conflict from the people who were actively involved in the conflict. The interviewees comprised mainly of members of the CAAC, other members of ICAI knowledgeable about the subject, and audit partners of local and Big Four accounting firms who were affected by the entry of the multinational accounting firms. The interviews were used to obtain information which may not be effectively captured in the archival documents. The interviewees were asked to think about additional materials that might be available, including additional archival documents. The interviewees were also asked to discuss the contexts of the committees on which they served and of the reports, if any, they authored. Finally, the interviewees were asked to validate the chronology of events obtained from the archival documents. The interviews were started by describing the objective of the research and by introducing an informed consent form, which both the interviewer and the interviewee needed to sign. In particular, the interviewees were asked for permission to tape the interview, while emphasizing that complete anonymity would be provided to them and their current employing organization. Also, the interviewees were told that they would have the opportunity to subsequently verify the accuracy of the transcript and add changes that they feel might be needed to make them comfortable with what they said during the interview. Appendix B provides the occupations of the interviewees, and the time the interviews were conducted.

The Indian Accounting Profession

In India, prior to 1949, the profession of accountancy in India was controlled and regulated by the government, which subsequently vested the regulatory power in the ICAI through an enactment in the parliament in 1949. The enactment was The Chartered Accountant Act of 1949 for the regulation of the profession of accounting in India. This act established a regulatory body for the Indian accounting profession in the form of the ICAI. The ICAI is



responsible for conducting examinations, development of uniform standards of practice and the strategic development of the profession. It is also responsible for licensing, certification, training, and discipline (CAAC, 2002). Table 9 below provides a detailed timeline of events related to the birth of the ICAI to the present day.



Table 9. Timeline of events for the ICAI

1857 The first ever Companies Act in India legislated. 1866 Law relating to maintenance of accounts and audit thereof introduced. Formal qualification as auditor now required. New Companies Act enacted. Books of accounts to be maintained specified. Formal qualification to act as auditor named. A Certificate from the local government to be held in order to act as auditor. An unrestricted Certificate entitled a person to act as auditor only within the Province concerned and in the languages specified in the certificate. 1918 Government Diploma in Accounting (GDA) launched in Bombay. On completion of articleship of three years under an approved accountant and passing the Qualifying Examination the candidate would become eligible for the grant of an Unrestricted Certificate. 1920 The issue of Restricted Certificates discontinued. 1927 Society of Auditors founded in Madras. 1930 Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. 1932 First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. 1933 First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1943 GDA abolished. 1949 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountants act was brought into effect on 1st July. The Institute of Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	Year	Event				
1913 New Companies Act enacted.	1857	The first ever Companies Act in India legislated.				
1913 New Companies Act enacted.	1866					
Books of accounts to be maintained specified. Formal qualification to act as auditor named. A Certificate from the local government to be held in order to act as auditor. An unrestricted Certificate entitled a person to act as auditor throughout British India. A Restricted Certificate entitled inim to act as auditor only within the Province concerned and in the languages specified in the certificate. 1918 Government Diploma in Accounting (GDA) launched in Bombay. On completion of articleship of three years under an approved accountant and passing the Qualifying Examination the candidate would become eligible for the grant of an Unrestricted Certificate. 1920 The issue of Restricted Certificates discontinued. 1927 Society of Auditors founded in Madras. Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1948 GDA abolished. Expert Committee formed to examine the scheme of an autonomous association of accountants in India. The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants of India is born.		Formal qualification as auditor now required.				
Formal qualification to act as auditor named. A Certificate from the local government to be held in order to act as auditor. An unrestricted Certificate entitled a person to act as auditor throughout British India. A Restricted Certificate entitled him to act as auditor only within the Province concerned and in the languages specified in the certificate. 1918 Government Diploma in Accounting (GDA) launched in Bombay. On completion of articleship of three years under an approved accountant and passing the Qualifying Examination the candidate would become eligible for the grant of an Unrestricted Certificate. 1920 The issue of Restricted Certificates discontinued. 1921 Society of Auditors founded in Madras. 1930 Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. 1932 First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. 1933 First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India born.	1913	New Companies Act enacted.				
to be held in order to act as auditor. An unrestricted Certificate entitled a person to act as auditor throughout British India. A Restricted Certificate entitled him to act as auditor only within the Province concerned and in the languages specified in the certificate. 1918 Government Diploma in Accounting (GDA) launched in Bombay. On completion of articleship of three years under an approved accountant and passing the Qualifying Examination the candidate would become eligible for the grant of an Unrestricted Certificate. 1920 The issue of Restricted Certificates discontinued. 1927 Society of Auditors founded in Madras. 1930 Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. 1932 First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. 1933 First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1936 GDA abolished. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.		Books of accounts to be maintained specified.				
articleship of three years under an approved accountant and passing the Qualifying Examination the candidate would become eligible for the grant of an Unrestricted Certificate. 1920 The issue of Restricted Certificates discontinued. 1927 Society of Auditors founded in Madras. 1930 Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. 1932 First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. 1933 First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.		to be held in order to act as auditor. An unrestricted Certificate entitled a person to act as auditor throughout British India. A Restricted Certificate entitled him to act as auditor only within the Province concerned and in the languages specified in the certificate.				
1927 Society of Auditors founded in Madras. 1930 Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. 1932 First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. 1933 First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1918	articleship of three years under an approved accountant and passing the Qualifying Examination the candidate would become eligible for the grant of an Unrestricted				
Register of Accountants (RA) to be maintained by the Government of India to exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). 1930 The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. 1932 First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. 1933 First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1940 GDA abolished. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1920	The issue of Restricted Certificates discontinued.				
exercise control over the members in practice. Those whose names found entry here were called Registered Accountants (RA). The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. The first Final Examination was held. GDAs exempted from taking the test. GDA abolished. Expert Committee formed to examine the scheme of an autonomous association of accountants in India. The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1927	Society of Auditors founded in Madras.				
The Governor General in Council replaced the local government as the statutory authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. The first Final Examination was held. GDAs exempted from taking the test. GDA abolished. Expert Committee formed to examine the scheme of an autonomous association of accountants in India. The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1930	exercise control over the members in practice. Those whose names found entry here				
authority to grant certificates to persons entitling them to act as auditors. Auditors allowed to practice throughout India. First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. The first Final Examination was held. GDAs exempted from taking the test. GDA abolished. Expert Committee formed to examine the scheme of an autonomous association of accountants in India. The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1930					
First Accountancy Board formed. The Board was to advise the Governor General in Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. The first Final Examination was held. GDAs exempted from taking the test. GDA abolished. Expert Committee formed to examine the scheme of an autonomous association of accountants in India. The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.						
Council on matters relating to accountancy and to assist him in maintaining standards of qualification and conduct required of auditors. First examination held by the Indian Accountancy Board. GDAs exempted from taking the test. The first Final Examination was held. GDAs exempted from taking the test. GDA abolished. Expert Committee formed to examine the scheme of an autonomous association of accountants in India. The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.		Auditors allowed to practice throughout India.				
taking the test. 1935 The first Final Examination was held. GDAs exempted from taking the test. 1943 GDA abolished. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1932	Council on matters relating to accountancy and to assist him in maintaining standards				
1943 GDA abolished. 1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1933					
1948 Expert Committee formed to examine the scheme of an autonomous association of accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1935	The first Final Examination was held. GDAs exempted from taking the test.				
accountants in India. 1949 The Chartered Accountants Act, 1949 was passed on 1st May. The term Chartered Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1943	GDA abolished.				
Accountant came to be used in place of Indian Registered Accountants. Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1948	accountants in India.				
Chartered Accountants Act was brought into effect on 1st July. The Institute of Chartered Accountants of India is born.	1949	<u> </u>				
		Chartered Accountants Act was brought into effect on 1st July. The Institute of				
	1999					

The ICAI has its headquarters at New Delhi with five regional offices at Mumbai,

Chennai, Kanpur, Kolkata, New Delhi and 98 branches spread all over the country. In addition, it



has also set up 11 chapters outside India. The total membership of the ICAI is about 115,000 and over 250,000 students are pursuing the Chartered Accountancy course. Out of the total members, nearly 70% are in practice (www.icai.org). In addition to regulating the profession, the ICAI is a national standard setting body in India and the National Accounting Standards are given statutory recognition under various statutes. The Indian Accounting Standards have almost been harmonized with the International Accounting Standards. In addition, the ICAI also issues from time to time various guidance notes, monographs, etc. for the benefit of its members on various new issues. The ICAI also issues Auditing & Assurance Standards (AAS) which codify the audit practices to be followed by the members of the ICAI, whenever an audit is carried out. The AASs' are mandatory in nature. The ICAI has issued 30 AASs' so far which meet international benchmarks and expectations.

The affairs of the Institute are managed by the Council in accordance with the provisions of the Chartered Accountant Act of 1949 and the Chartered Accountants Regulation of 1988.

The Council consists of 30 members of whom 24 are elected by the members and remaining 6 are nominated by the Central Government to represent the Comptroller and Auditor General of India, Central Board of Direct Taxes, Department of Company Affairs and other stakeholders.

The Council functions through three Standing Committees and sixteen Non Standing

Committees for carrying out the provisions of the Chartered Accountant Act of 1949

(www.icai.org). The decision making process in the council is structured so that all 30 members have to consensually agree on any decision (www.icai.org). The Indian accounting firms are subject to rules and regulations made under the Chartered Accountant Act of 1949.

One of the significant restraints that the ICAI places on its members is that Indian accounting firms cannot advertise, whether in India or abroad (CAAC, 2002). They cannot



canvass directly or indirectly for professional assignments nor can they make presentations to prospective clients. Further, Indian accounting firms cannot have non Chartered Accountants (hereafter, CAs') as partners nor can they have any profit sharing arrangement with non CAs' (CAAC, 2002). The Indian accounting profession is mainly comprised of small and medium sized firms, the number of firms with five or more partners being only about 375 in 2001 out of a total of 42,339 firms. The table below gives the distribution of firms by size in 2001 (CAAC, 2002).

Table 10. Size distribution of CA firms as at 01/04/2001

No of Partners	No of Firms
2	7161
3	2104
4	796
5	375
6	305
7	206
8	101
9	61
10	34
>10	52

The case of India: Entry of multinational accounting firms

The Foreign Exchange crisis in 1990: Liberalization of the Indian Economy

The age of liberalism, a period of relatively open global markets, ended after the Great Depression of the 1930s (Polanyi, 1944). In the post- World War II period, it has been held that nation states instituted Keynesian economic policies and political controls over the international movement of financial capital (Schor, 1992). In the few decades after independence in 1947, India adopted a development strategy which was one of national self-sufficiency and stressed the importance of government regulation of the economy. Cerra et al characterized it as "both inward looking and highly interventionist, consisting of import protection, complex licensing requirements, and pervasive government intervention in financial intermediation and substantial public ownership of heavy industry" (2000:3). The trade regime in India was characterized by high nominal tariffs and pervasive non-tariff barriers, including a complex import licensing system, an "actual user" policy that restricted imports by intermediaries, restrictions of certain exports and imports to the public sector, phased manufacturing programs that mandated progressive import substitution, and government purchase preferences for domestic producers (Cerra et al., 2000). The white paper reiterates this mindset in the Indian economy:

The socialist psyche was so much internalized in politics that in the late 1960s and early 1970s, anyone who disagreed with the socialist model was labeled as anti-poor and pro-rich, pro-America and even as CIA agents! Those who differed from socialist ideals were ostracized as political untouchables. With the result that, every political party was compelled to file an undertaking expressing faith in the ideology of socialism, to be registered as a political party eligible to contest elections. In fact, even the constitution of India was amended to define India as a socialist state. Even the property rights listed as unbridgeable fundamental rights were deleted and relegated to the status of mere legal rights. The effect of this is to facilitate even expropriation of property with or without adequate compensation. (CAAC, 2002: 6)

The Indian accounting profession operated in the socialist environment prevalent prior to the 1980s. Up to 1949, accounting was controlled and regulated by the Government, which



eventually vested the regulatory power in the ICAI through an enactment in the parliament in that year (CAAC, 2002). The structure of the Indian accounting profession was a reflection of the structure of the Indian economy. As the partner of a local accounting firm points out, the Monopolistic and Restrictive Trade Practices Act, 1969 (MRTP Act) effectively prohibited the formation of large companies in all the sectors of the economy and so the Indian economy was comprised of small and medium-sized companies. In a similar manner, restrictions were placed on the maximum number of partners per accounting firm. Consequently, the size of an accounting firm was limited to less than 20 partners. Further, accounting firms were also prohibited from providing multi-disciplinary range of services and from entering into profitsharing arrangements with non-members of the ICAI. In the metropolitan centers, audit work constituted the bulk of the work of medium sized accounting firms, while taxation services were the mainstay of small sized accounting firms. The services rendered by the accounting firms included attestation services, taxation services and regulatory compliance services, which largely meant handling state regulations. The pre-dominance of audit and assurance services in the service portfolio of local accounting firms is captured in the quote below by the partner of a local accounting firm:

There was not much need for consulting services due to the nature of the market as the Indian economy was not only insulated from the global market but also had serious restraints on local market forces. On the consultancy side, which had not really grown in the absence of the market drive to the economy, handling the government, its regulations and instrumentalities for the private sector was the core part of the practice (Local firm partner).

In the restrictive regime prevalent in the Indian environment, as a partner of a Big Four firm puts it, most of the Indian companies were family-owned and as a result, much of the professional accounting work was founded on "relation-based approach as distinguished from a pure contract-based model, which is the distinctive feature of a marked driven economy". The



white paper emphasized that the Indian accounting firms operated in a protected environment, where they faced no competition from foreign accounting firms and they were also not allowed to operate in a foreign country. They did not have to adopt any strategies to maintain their monopoly over the practice of accountancy and therefore did not acquire any skills to create such strategies in the event that the market was opened which is what happened in the late 1980s and the early 1990s (CAAC, 2002).

In the 1980s and 1990s, it has been held that most of the nation states dismantled capital controls, bringing in a neoliberal period where once again capital is free to roam the world relatively unconstrained by national controls (Kapstein, 1994). This US agenda known as the "Washington Consensus" called not only for free trade and elimination of capital controls, but also for deregulation and economic restructuring. It was argued that IMF-imposed structural adjustment programs were used to coerce weak nation states such as India to abandon domestic priorities in favor of open borders and investor-friendly economies (Stiglitz, 2002).

In such a free trade environment prevalent in the world, over the period 1990-1992, public reports of India's financial problems generated rumors of an impending foreign exchange crisis and the consequent probability of default on its borrowings from international markets and banks (Clark et al., 2003). The situation was further compounded by a variety of macroeconomic imbalances - the increase in oil prices due to the conflict in the Middle East in 1990, the reduction in remittances from Indian workers in the same region, and the reduced demand of important trading partners, as well as political uncertainty (Clark et al., 2003). To deal with the problem of shortage of foreign exchange reserves leading to non payment of external debt, the government of India requested a Stand-By Arrangement from the IMF in August 1991. The reform package from the IMF was conditional on an adjustment program featuring



macroeconomic stabilization and structural reforms. The reforms were mainly in the areas of industrial and import licenses, the financial sector, the tax system, and trade policy. On trade policy, benchmarks for the first review of the Stand-By Arrangement included a reduction in the level and dispersion of tariffs, a removal of quantitative restrictions on imported inputs and capital goods for export production, and elimination of public sector monopoly on imports of all items except petroleum, edible oils, and fertilizers and certain items earmarked for health and security reasons (Clark et al., 2003).

Post Liberalization: The Entry Process

It is in the backdrop of these sweeping reforms, that I analyze the process of entry of the multinational accounting firms into India. In the process of negotiating the "bailout package" with the IMF, the multinational accounting firms observed an excellent opportunity to enter the Indian market for consulting services (CAAC, 2002). The firms entered into India as consultants through the stipulations laid out by the IMF in the reforms package offered to the Indian government. The argument was that the multinational accounting firms were needed to facilitate the free flow of the foreign direct investments into India in the liberalized regime by the multinational corporations and investment bankers, who would only be comfortable with their own consultants as advisors about the "best ways "to invest in India (CAAC, 2002:30; www.icai.org). Faced with a foreign exchange crisis and possibility of default on its foreign loans, the Reserve Bank of India issued special licenses to the multinational accounting firms to provide consulting services in India. In the early 1990s, the Foreign Exchange Regulation Act, 1947 (FERA) was in force in India to control the outflow of foreign exchange reserves. Under the provisions of the act, foreign firms could enter the Indian market only after obtaining licenses issued by the Reserve Bank of India. The legal provision under which the multinational



accounting firms obtained licenses to operate in India is contained in Section 29 of the FERA act, which was eventually abolished in 2002.

Post liberalization, the consulting services market provided new opportunities for the Indian accounting profession in terms of providing services to the multinational companies investing in India. However, the white paper reiterates that the entry of the multinational accounting firms in the consulting services market gave them a distinct competitive edge over the local accounting firms. One of the authors of the white paper expanded on the reasons for the lack of level playing field for the local accounting firms in the new environment. According to him, while the liberalization program allowed for foreign direct investment in the other sectors of the economy, the tariff regime was not entirely dismantled which had the effect of protecting the local companies from competition by the multinational companies. The Indian currency was also devaluated in a phased manner, which allowed the local manufacturers to prepare for competition from foreign operators. However, this kind of phased liberalization was not carried out in the accounting profession. For instance, while the MRTP act was abolished allowing the formation of large Indian companies; the restriction regarding the maximum number of partners per accounting firm (20) was not removed. Thus, while the size of the local accounting firms mirrored the size of the Indian companies till the advent of the liberalization program, it was no longer the case after 1992. While the Indian companies began growing in size due to the opening up of the economy and the advantages of consolidation, the structure of the local accounting firms remained the same. As one of the interviewees who is the partner of a Big Four firm pointed out:

It was very difficult for these small local firms to meet the demands of the large scale expansions of businesses and provide quality services, which only the Big Four firms could handle! This paradigm virtually robbed the Indian accounting firms of their



competitiveness, so very crucial to take on the Big Four accounting firms in India (Big Four firm partner).

This fact and the reason for the success of the multinational accounting firms in the consulting services market is also evidenced in the reference below:

This was made possible because the MAFs managed to come to India stealthily by coining their connections in the international financial system and using the financial crisis which India faced. Thus the entire consulting services market came to be monopolized by the MAFs (CAAC, 2002: 33).

This reforms package appears to be an obvious evidence of agency and intent in the process of economic globalization. The administrative apparatus of institutions such as the IMF makes them a significant institutional actor in their own right (Arnold 2005). However, the IMF does not act on its own. The institution is heavily influenced by the interests of major nation states such as the United States and members of the European Union, and transnational firms with operational bases within the borders of the major nation states. The dominant role of the major powers within the IMF, the WTO, and the World Banks has been documented (Stiglitz, 2002). In turn, it has been held that multinational companies and industry lobbies have substantial levels of influence over the international trade agendas of their nations (Arnold, 2005). As the Held (1991) framework has argued, multinational corporations are not 'free floating agents', on the other hand, they maintain strong links with the countries of their origin(Cooper et al. 1998), and have the "potential to mobilize their national governments to influence the rules of the game to their own advantage" (Caramanis, 2002: 403).

A second and more important theme of the white paper was that the Indian accounting profession and the ICAI were not informed about the entry of the multinational accounting firms into India, nor were they aware of the mode of entry into India. The multinational accounting firms entered India through the "floodgate of reform measures initiated by the government of



India" (CAAC, 2002: 31). There was no debate about the advantages and disadvantages of the entry of the multinational accounting firms in the Indian market at any forum (CAAC, 2002; www.icai.org). This sentiment is captured in the following reference in the white paper:

The entry was stealthy. Even ICAI was not consulted; perhaps it knew of the decision only in newspapers. While it is true that the initial 'mistake' could probably be attributed to the RBI and the government in as much as ICAI was not even consulted before such opening up , the leadership of the profession has also failed in its part for its apparent apathy and indifference over the continued aberration , by not raising adequate objections. (CAAC, 2002: 31)

One of the interviewees, who is a director of a surrogate firm formed by a Big Four firm and in charge of audit practice in India, was not even aware of the consulting licenses granted to the multinational accounting firms by the Reserve Bank of India. This lack of knowledge is more surprising in light of the fact that the director in question was an ex-president of the ICAI. One of the authors of the white paper confirms the silent entry of the multinational accounting firm in the Indian market and attributes it to the impulsive reaction of the Indian government to the foreign exchange crisis.

True, the ICAI was not consulted at any stage as the central government and the Reserve Bank of India (the nation's Central Bank), which were working overtime to contain the foreign exchange crisis succumbed to the pressure of the IMF and the World Bank and the global financial pressure groups in ordering un-calibrated opening of the economy and did not deem it fit to consult the premier accounting body in any such decision making process. Opening of the consultancy field was regarded as the concomitant need to attract foreign direct investment. Since a national consensus had been built in favor of attracting foreign investment at any cost, and the opening of the consultancy sector was part of the agenda to attract foreign exchange through FDI, the opening of the consultancy sector also became integral to the emergency measures taken to correct the imbalances in the exchange regime (White Paper author).

It was in the late 1990s that some members of the ICAI started questioning the "west-centric model of globalization of the accounting profession in India" (CAAC, 2002:7). The process slowly evolved by extensive discussions, meetings and interactions within the Indian



accounting profession. The eventual result was the birth of the CAAC at Chennai, India and the publication of the white paper. The first convention of the CAAC was held on September 1, 2002, when over four hundred CAs' gathered together to discuss the implications of the entry of the multinational accounting firms in the Indian market. The members of the CAAC interacted with businessmen and policy makers as well as the media all over India and found an "amazing lack of awareness about the illicit entry of the multinational accounting firms" (CAAC, 2002:7). The members of the CAAC felt that it was necessary to shed light on the causes and consequences of the entry of the multinational accounting firms on the Indian accounting profession. The CAAC was chaired by Mr. B.S. Raghavan, a retired civil servant, who was not a CA and so it was felt that the content of the white paper would not be subject to any unintended bias. The other members of the CAAC were all CAS. It is important to note here that the ICAI was neither involved in the creation of the CAAC nor in the publication of the white paper.

Not only was the Indian accounting profession not informed about the entry of the multinational accounting firms in India, no reciprocal right was conceded to the Indian accounting firms to provide accounting services in the USA or the UK, the countries widely considered as the places of origin for the Big Four accounting firms (Caramanis, 2002). This is expected to have an adverse impact on the Indian accounting firms in the General Agreement on Trade in Services (hereafter, GATS) negotiations under the auspices of the WTO. The GATS is an existing trade agreement, whose objective is to open the borders of the WTO member nations to trade in all types of services, including accounting and auditing (www.wto.org). It not only covers all types of services; it also applies to all possible modes of delivering services². The United States has made commitments at the GATS negotiations to open its borders to all modes

² GATS has four possible modes of delivering services, including: 1. cross-border delivery, 2. consumption abroad, 3. commercial presence, and 4. movement of natural person. For technical definitions of all four modes, see Arnold(2005)

of delivery except for Mode 4, which talks about the movement of persons where individuals from one country move temporarily to another country to deliver accounting or auditing services (WTO, GATS/SC/90, 1994). Further, the United States has also written some limitations into its GATS commitments in order to protect specific state-level accountancy laws that would otherwise be GATS illegal, such as residence requirements for CPA licenses and laws that limited ownership of accounting firms to persons licensed as accountants. This action is seen by the white paper as a "deliberate game plan of the west to maintain their control of the financial world" (CAAC, 2002:31).

In the Uruguay Round of the GATS negotiations (1986-1994), nation-states undertook limited commitments to lock in the status quo (Arnold, 2005). The effect of locking in existing levels of liberalization is that once a company undertakes commitments in the accounting sector, they are binding and cannot be altered without substantial sanctions (Arnold, 2005). Thus, the Indian accounting profession had no leverage in the negotiations in the WTO as the Big Four accounting firms had already entered the Indian market and did not have to offer any reciprocal benefits to the Indian accountants. The reason is that the GATS legal framework contains provisions that make it difficult for nations to modify or withdraw commitments once they are undertaken. The framework contains provisions to the effect that a WTO member may modify or withdraw commitments, but it has to then "compensate" any WTO member who is affected by the modification or withdrawal of the commitment (Arnold, 2005). Accordingly, India is effectively bound to the liberalization and privatization that took place during the 1980s and 1990s and the Indian government may find it very costly to reinstate restrictions on market access or policies favoring local accounting firms. For instance, when local Greek auditors attempted to regain control of the market after it had been liberalized in 1992, the United States



in collusion with the international accounting firms put intense political pressure to successfully block the attempt (Caramanis, 2002). Given the broad consensus that it may take another four or five years for the GATS to become effective (Arnold, 2005), it means that even if the Big Four accounting firms eventually give reciprocal rights to the Indian accounting firms, the Big Four accounting firms have an early entry advantage of around fifteen years over the Indian accounting firms in the Indian market.

Further, the WTO conducted a survey in 1998 to analyze the limitations faced by developing countries in the GATS negotiations (WTO, 1998). According to this survey, in most of the developing member countries, including India, incorporation as a corporate body is prohibited and partnership is the only collective form of practice allowed. This automatically puts the local accounting firms, which cannot have more than twenty partners and cannot organize themselves as a corporate entity at a disadvantage as compared to the multinational accounting firms, which don't have this restriction placed on them.

Post Entry: Entry in the traditional auditing and assurance market

In the post-liberalization regime, there were a host of new opportunities for the local accounting profession in the area of providing consulting services for the multinational corporations. However, the white paper argued that the local accounting firms could not fully utilize these opportunities due to the entry of the multinational accounting firms, which were given unlimited access to the market by the sweeping reforms in 1991 (CAAC, 2002). The white paper emphasized that if the entry of the MAFs was done in a phased manner; it would have given the Indian accountants ample opportunities to gain the requisite knowledge, skills and expertise to take advantage of the opportunities afforded by the trend of globalization.



The Big Four accounting firms could not directly enter the audit and assurance market due to certain provisions of the Indian Companies Act of 1956 and the Indian Chartered Accountant Act of 1949. The Chartered Accountant Act of 1949 prohibits non-CAs' from rendering audit and attestation services. Further, Section 11(2) of the Indian Companies Act of 1956 restricts the number of partners in a partnership firm to 20 partners. There are also restrictions on the acceptance of the number of audits that an Indian accounting firm may accept. The Indian Companies Act requires that the audit report and the balance sheet have to be signed by a partner of the firm in his own name and not in the name of the firm. This implies that the responsibility of the partner signing the documents is personal and unlimited.

Some of the other restrictions imposed by the ICAI on its members are that an Indian accounting firm cannot advertise, whether in India or abroad. They cannot canvass directly or indirectly for professional assignments, nor can they make presentations to prospective clients. There are rigid restrictions on the content of the websites of the accounting firms, which are laid out in the Guidance Note issued by the ICAI (www. icai.org). Indian accounting firms cannot have non-CAs' as partners. Nor can the Indian CAs' or accounting firms have any profit-sharing arrangement with non-CAs'. Thus it is not possible for Indian CAs' to structure multi disciplinary practice to offer a one stop professional facility to clients.

In the light of such rigid restrictions, the Big Four accounting firms entered into an arrangement with a few large Indian accounting firms, turning them into "surrogate firms" (CAAC, 2002:34). Each of the Big Four accounting firms has more than one surrogate firm. Most of the surrogate firms are well established large Indian accounting firms in the traditional areas of audit and assurance services (www. icai.org). The list of the surrogate accounting firms of the Big Four accounting firms is provided in Table 2.



This arrangement enabled the multinational accounting firms to enter the audit and assurance services market, so that the partner of the surrogate firm acted in two capacities, first as partner of the surrogate firm providing audit services and second as the director of the subsidiary of the multinational accounting firm, which was licensed to provide consulting services. Even though there were two entities, one an audit entity and another a consulting entity, both the entities shared the same premises, the same staff, the same clients and even the same website and email. With the help of the surrogate firms mentioned in the table above, the Big Four accounting firms started rendering attest and assurance services, which could only be provided by local accounting firms incorporated by CAs' licensed by the ICAI. As the consultancy arms of the surrogate firms are incorporated as corporate entities, under the Indian Companies Act, 1956, there is no prohibition for a person to be a director of the consulting entity and still remain a full time practicing partner of the surrogate firm to provide the attest functions. Thus one of the interviewees is actually both the director of the subsidiary consulting entity of a Big Four accounting firm and also a partner of the surrogate firm formed by the same Big Four accounting firm. As one of the authors of the white paper points out and the partner of a Big Four firm confirms the fact:

The partner of the local audit firm would also be the director/representative of the linked Big Four consultancy, so that one visiting card will identify him as the representative of the Big Four entity and the other visiting card will identify him as the partner of the audit firm (White paper author).

The profit sharing arrangement between the surrogate firms and the multinational accounting firms is not publicly known according to one of the authors of the white paper and an analysis of the archival documents available on the ICAI website. However, I obtained this information from the interviews conducted with the partners of the surrogate firms. The client of



the surrogate firm has the option to choose from a "menu" of firms, which are a part of the surrogate firm, for providing services. The surrogate firm then provides the services to the client using the same office space, staff and technical resources available to all the firms forming the surrogate firm. Since the multinational accounting firm cannot share profits with the local accounting firm consisting of CAs', the multinational accounting firm charges the local accounting firm fees in the form of "technical consultancy" or "technology fees" or "training fees", which are in turn, reimbursed by the client. Thus, in technical terms, there is no violation of the regulations contained in any of the Indian acts regulating the Indian accounting profession.

It is obvious that there is no free lunch between the concerned MAF and the surrogate firm. It is certain that there is consideration, but no one knows how the consideration is factored in. There are surmises that the consideration could be disguised in the form technical consultancy, technology fees, training fee etc. As the consultancy arms of Big 4 are in the corporate forms, as per the prevailing law, there is no bar for the same person to be a director (not whole time or managing) of the consulting private limited company, and yet remain a full time practicing professional (as a partner in a surrogate firm) to do the attest functions (White paper author).

The surrogate arrangement provided benefits to both the Big Four accounting firms as well as the few local accounting firms, which were involved in the arrangement. As the traditional areas of practice were changing with the entry of foreign direct investment, the entry of foreign institutional investors and their investments in local companies, in addition to the open partisan support of the world bodies such as the IMF and the World Bank for the multinational accounting firms, the local Indian accounting firms too "felt insecure about their capacity to retain their position, and therefore, many of them began thinking in terms of becoming their affiliates or surrogates to retain the very work they were handling and to access new work through the MAFs" (CAAC, 2002: 34). Besides gaining an entry in the traditional audit services market, the Big Four accounting firms had one more reason to enter into an arrangement with the large Indian accounting firms. The council members of the ICAI appointed by the members of



the ICAI were represented to a great degree by the partners of the surrogate firms (www.icai.org/members). Although the surrogate firms constituted less than 2% of the entire population of local accounting firms, many of the ex-presidents and current presidents were partners of the surrogate firms. For instance, the partners of two of the Big Four accounting firms currently in charge of operations in India were presidents of the council of the ICAI for the years 1998 and 1993 (www.icai.org). Therefore, the arrangement enabled the Big Four accounting firms to gain some leverage in the decision making of the ICAI.

While the Indian accounting firms are subject to the strict restrictions mentioned above such as prohibition of canvassing or advertising their services, the multinational accounting firms began to openly advertise and build their brands in their own names. These promotion activities ranged from holding cricket matches to high cost advertisements in the media to events such as instituting and giving out Business Leadership and Entrepreneur awards, in order to gain access to the high yielding corporate and financial market for professional work(CAAC, 2002). The ICAI clearly expressed its displeasure at this activity, as is evidenced from a report published in "The Chartered Accountant", the official organ of the ICAI, which notes that:

The Government should review the alternative route of entry of accounting firms in India in the name of management consulting firm, and circumvention of the law of the land taking place directly and indirectly by performing accounting services by them(2000:16)

In 2003, a panel appointed by the central council of the ICAI recommended that the multinational accounting firms be prohibited from offering accounting and consulting services under their current names (ICAI, 2003). The panel recommended that the ICAI should not allow multinational accounting firms offering consulting or accounting services to bear a name that is similar to that of an Indian or foreign accounting firm. Acknowledging that the ICAI was not vested with jurisdiction over multinational accounting firms, the panel suggested that the ICAI



should make a representation to the department of company affairs on the adverse consequences of allowing the Big Four accounting firms to retain their names for providing management consultancy services. The panel suggested that the department of company affairs can then instruct the Registrar of Companies to prohibit the use of such names. The panel also suggested that the 'surrogate' firms should not indulge in "co-branding and attest functions directly or indirectly" (ICAI, 2003:1). It is apparent that the ICAI failed to enforce the recommendation of the panel as the Big Four accounting firms to the present date continue to provide consulting and accounting services under their own names (www.icai.org).

Support of the Indian Government and the Indian Financial Institutions

To compound the woes of the already beleaguered local accounting firms, the white paper argued that the local official regulators, banks, and even governmental institutions were responsible for further undermining the efforts of the local accountants to establish some foothold in the professional market for services, by actively patronizing multinational accounting firms (CAAC, 2002). This patronage compounded by the enormous influence of the multinational accounting firms in the international arena, it is argued, adversely impacted any hopes for the Indian accountants to protect their jurisdiction of practice. The authors of the white paper unanimously emphasized this patronage of the multinational accounting firms by the local regulators, specifically the ICAI. They emphasized that while the white paper generated tremendous interest in the media and in the government, the ICAI "did nothing about it and it remained and even today, it maintains a deafening silence on the white paper" (White paper author). On the contrary, the ICAI published the vision document at around the same time as the white paper, which openly legitimated the presence of the multinational accounting firms in



India and exhorted the local accounting firms to accept this reality (ICAI, 2004). The tone in the vision document is evidence by the reference below:

The Indian Chartered Accountancy Profession must recognize the changes in Economy/Business environments, globalization of business and competitive pressures... and must recognize the path to success by adapting to the changes, knowledge management and acquiring skills to work with future environment influenced by technological and other changes. It must recognize the Institute's role as proactive, innovative and flexible and the need to be known as a World Class Advisor (ICAI, 2000: 1)

The reason for the support of the entry of the multinational accounting firms in the Indian market by the ICAI is provided by one of the authors of the white paper. According to him, the ICAI was in a "catch-22 situation". The ICAI cannot openly oppose the entry of the multinational accounting firms in the Indian market, unless it is willing to discipline the surrogate firms who are violating the rules of the ICAI, as the partners of these surrogate firms are influential members on the decision making committees of the ICAI. On the other hand, if the ICAI does not acknowledge the entry of the multinational accounting firms, it appears to be weak in the eyes of the local accounting firms who have been affected by the success of the multinational accounting firms in the audit and assurance market. So the only option available to the ICAI is to legitimize the actions of the multinational accounting firms by framing their entry as an event which is in line with the changes in the global environment and the core values of the profession.

Besides the ICAI, the local governmental authorities were providing tacit support to the multinational accounting firms to the detriment of the local accounting profession. For instance, when the divestment program of the Oil companies, which were in the public sector, was postponed by the government in 2002 for lack of political consensus, the international rating agencies immediately downgraded the investment rating of India. This rating change happened



in spite of the fact that the Indian economy was actually booming and the exports were increasing at an unprecedented pace (Topalova, 2004, IMF working paper). Further, the State Bank of India issued an advertisement inviting bids from accounting firms, in India and abroad, but the stipulation was that the size of the firm in terms of assets had to be around Rs. 500 billions. The State Bank of India is aware of the fact that there is no local accounting firm, which has assets over Rs. 500 billions. (CAAC, 2002). The patronage enjoyed by the multinational accounting firms from the Indian government is argued to range from lucrative assignments in the planning commission, advisory role disinvestments, and advisory role in implementation of new taxation structures. Table 11, which shows the list of the public sector units that, hired the multinational accounting firms as advisors provides evidence of this patronage.



Table 11. List of Disinvestments with Foreign and Indian advisors

Number	Name of the PSU	Advisor(Foreign)	Advisor(Indian)
1	Air-India	JP Morgan Stanley	
2	CMC Limited	KPMG	
3	Hindustan Copper	Sumitomo Bank	IDBI
4	Hindustan Insecticide		A F Ferguson(*)
5	Hindustan Organic Chemicals		A F Ferguson
6	HTL	KPMG	
7	Hindustan Zinc Ltd	BNP Paribas	
8	Indian Airlines	ANZ Grindlays	IDBI
9	IBP Limited	HSBC Securities	
10	HPCL	Warburg Dillion Read	
11	ITDC	Lazard	
12	Madras Fertilizers	Bank of America	
13	National Fertilizers Limited	Rabo Finance Limited	
14	Pradeep Phosphate Limited	Deloitte and Touche	
15	Sponge Iron India Limited		A. F. Ferguson
16	VSNL	CS First Boston	SBI Caps
17	Bharat Heavy Plates		S. B. Billimoria(*)
18	Bharat pumps		S.B. Billimoria
19	Hindustan Cable Limited		ICICI
20	Hindustan Salts		SBI Caps
21	Instrumentation Limited		IDBI
22	Jesop and Co		A. F. Ferguson
23	NEPA		SBI Caps
24	Scooters India	PWC	
25	Tungabhadra Steel		IDBI

In one of the interviews, an author of the white paper attributes this patronage to the

"brand" reputation and soliciting activities of the multinational accounting firms.

The Indian banking and public sector fell prey to the attractive business models of the MAFs. To illustrate, there have been numerous instances where advertisements calling for empanelling CAs would be tailored to suit exclusively the MAFs. This was done repeatedly by some Banks and other financial Institutions when the MAFs cannot be legally present in India was overlooked by everyone including the ICAI! With the result the national firms were excluded by tender conditions. The same tender conditions could have made it obligatory for an Indian CA firm to associate itself with a MAF if experience of the kind which the MAF was only known to possess. This has happened in the case of the telecom field where the government tender used to insist that only Indian companies should bid but it



must have a foreign investor who has a certain experience in telecom operations. But here in the case of the CAs the requirement was not to promote national CA firms, but to destroy them (White paper author).

Additional evidence of lack of level playing field for the Indian accounting firms is the recent change in US regulations. The committee preparing the white paper discovered that the US regulations have been modified to stipulate that the companies listed in the US could only be audited by an audit firm registered in the US and the country where the auditors practice (CAAC, 2002). The intended effect of this regulation is to force countries such as India, which has not allowed multinational accounting firms to directly register in India, to do the same. The reason is that if India does not register the multinational accounting firms in India, then neither the multinational accounting firms nor their surrogates can certify the accounts of the Indian companies listed in the USA. Thus, there are two scenarios. Either, the surrogates of the multinational accounting firms have to be registered in the USA or the multinational accounting firms have to be registered in India. Given the reluctance of the US state bodies to grant registration to foreign accounting firms and the tremendous influence enjoyed by the multinational accounting firms with the Indian government as evidenced in this paper, it is more likely that the multinational accounting firms will be allowed to register in India.

The white paper further cites the example of a government run development financing institution, which put tremendous pressure on its borrowers to get their quarterly income statements certified by the multinational accounting firms. The assignments carried remuneration as high as Rs.15 to Rs.20 millions for four quarters. The borrowers had to follow the recommendations of the financial institution as they needed the financial support of the financial institution. The tacit support enjoyed by the multinational accounting firms from the local



government and regulating agencies in India is captured in the following reference in the white paper:

The Indian government has turned into the biggest promoter of the multinational accounting firms to the prejudice of the Indian accounting profession. The Financial Institution nominees on the boards of companies (borrowers) also began insisting on the assisted companies (borrowers) to appoint the multinational accounting firms as consultants and even as statutory auditors (CAAC, 2002: 42)

Finally, the culmination of the marginalization of the local accounting firms in India can be evidenced by the Chartered Accountant (Amendment) Act, 2006, which was notified by the Central Government of India in the Gazette of India dated 23rd march, 2006 (hereafter, the Amendment Act). Some of these changes are directly relevant to the entry of the multinational accounting firms in the Indian market. The Act, for instance, has removed the restrictions on sharing of profits or fees by a member of the ICAI with certain categories of non-members, as prescribed by the ICAI from time to time, for the purpose of rendering professional services in and out of India. Further, the Act now permits a multi-disciplinary partnership, in or outside India, with certain categories of non-members, to be prescribed, from time to time, in the Regulations. It is also now permitted to secure any professional business through certain categories of non-members, to be prescribed, from time to time, in the Regulations. The prohibition of solicitation has been relaxed in the following circumstances: - if it occurs within the fraternity; or in the event of responding to tenders or enquiries issued by various users. Limited advertisement (subject to guidelines to be issued by the Council) is also now permitted. Thus, while the act clearly legitimizes the formation of the surrogate firms by the multinational accounting firms, the local accounting firms still cannot form a partnership consisting of more than twenty partners. It is difficult to conceive



any realistic challenge that can be posed by the local accounting firms with restrictions on their size, to the formidable presence of the multinational accounting firms.

Theoretical Insights

There are various theoretical insights to be derived from the evidence in this paper. First, there are very important lessons to be learned for the weak nation states in the world economy and politics just as Caramanis (2002) demonstrated in his paper from Greece's experiences during the debate on the liberalization of the statutory audit market in the 1990s. These nation states provide easy opportunities to multinational accounting firms for expansion, as the former are economically and politically too weak to resist significant international pressure (Caramanis 2002). The sheer size of the virtual financial market is hundred times the volume of real economic transactions in the world. This is evidenced by the fact that trading in derivatives which used to be in the region of \$18 billion a day in 1978 rose to over \$1 trillion a day in 1990 and to nearly \$2 trillions a day in 2002 as opposed to trading in real goods which is about \$700 millions a day in 2002(The Economic Times, 2002). The result is that this situation is likely to give enormous influence to the people who operate the financial market and the arbiters of its rules and instruments, which would be naturally the accounting industry. Therefore, no country which is not familiar with the mechanics of the global financial system can be expected to succeed in grasping the concept of globalization. This evidence in this paper emphasizes that the dangers posed by the forces of globalization to the local markets is not rooted in regressive nationalism, rather it is based on an appreciation of the fact that globalization of accounting markets has not been accompanied by a level playing field for the local accounting profession. While the multinational accounting firms have huge amount of resources at their disposal (Anissette, 2000; Chua & Poullaos, 1998), the local accounting firms are not likely to possess



either the resource base or the requisite bargaining power to compete with the multinational accounting firms. Tinker (2002) concluded that the Big Four accounting firms are an elite faction within the US accounting force, employing less than 1% of the total accounting workforce in the Unites States. The result is that local accountants in these weak nation-states are lured into seeking employment with the multinational accounting firms and multinational corporations with their attractive remuneration packages which cannot be offered by the local accounting firms. This would undoubtedly result in the elimination of local professional accounting firms and the domination of the accounting services market by the multinational accounting firms. Further, it has been argued that the interests of the Big Four accounting firms align more closely with those of other multinational corporations that with small, nationally based accounting firms (Arnold, 2005). This sentiment is echoed in the following quote by one of the authors of the white paper:

On the potentiality of the CAs, factors like the growing appetite of the MNCs for quality finance professional, the emerging BPO opportunities, and high salaries – aided by the narrowing space in independent 'practice' – have been luring the massive majority of the new generation CAs in to highly paid jobs even inside the country, with compensation levels comparable internationally. This would, no doubt, would result in dwindling of Indian professional accountancy firms, automatically strengthening only the MAFs whose illegitimate presence may even become a fait accompli in law. Thus the calculated and invasion of big four into India by leveraging on their global financial network and exploiting the financial misfortunes of India in the early 1990s – without and far ahead of the safeguards, reciprocity and national treatment which the WTO offers to national professions being available to India – has eroded the potentiality of the Indian CA profession which was defenseless against such invasion (White paper author).

Second, it has been held that licenses to perform statutory audits have been historically granted to only locally established entities that are either owned or controlled by locally accredited professionals (Arnold, 2005), which is also the situation in the Indian accounting profession. Only CAS are licensed to provide audit and attestation services and the number of partners in a partnership firm is restricted to 20. These rules led to the growth of a small industry



of sole proprietorships and partnerships and the organization of the Indian accounting profession as an autonomous self-governing guild. The evidence in this paper shows these national regulations are being increasingly dismantled or superseded by the practices of the Big Four accounting firms. The result is that small and medium-sized accounting firms are likely to become vulnerable to consolidation by large professional service conglomerates and independent practitioners are likely to be swept into the corporate workforce.

Third, the only defense which nation-states appear to have to counter the forces of globalization is to strengthen and galvanize the national accounting profession, not only at the local level, but also at a global level to protect and advance the larger national interest. It has been argued that the main responsibility for this task lies in the hands of the local professional associations such as the ICAI (Willmott, 1986). Greenwood et al. (2002) concluded that the role of professional associations is that of conservatism blended with reform. While these associations are responsible for the monitoring of prevailing institutional norms, they also have an added responsibility to protecting the interests of its members from external market pressures. The professional associations in post colonial societies, however, present a different picture. Johnson (1973) showed how the historical development of professions in the post colonial societies differed fundamentally from the western world. His research argued that rather than developing as autonomous, self-governing guilds, the professions in post-colonial societies have been and continue to be subject to different forms of social and occupational control from metropolitan centers, including that exercised by major British accountancy associations and more recently the US accountancy industry. Subsequent research has found patterns of unequal exchange from Anlgo-American influence over professional accountancy in weak nation states such as South Africa, Trinidad and Tobago, and Greece (Annisette, 2000; Caramanis, 2002;



Chua and Poullaos, 1998, 2002). This was evidenced by the weak response of the ICAI to the entry of the multinational firms into India. The ICAI was completely ignorant of the entry of the multinational accounting firms in India and found out about the same in the newspapers. Even after the entry of the multinational accounting firms, the ICAI made an unsuccessful attempt to prohibit the multinational accounting firms from providing services using their own names. The reluctance of the ICAI to put up a fight is evidenced in this particular reference in the white paper:

The ICAI, which does not normally get into controversies and is in fact a mechanism to discipline the profession rather than a mechanism to fight for, defend and protect the professionThe ICAI is also not structured to fight evil as it is a gentlemen's institution, not capable of handling the unequal war forced on the profession by the entry of multinational accounting firms.... (2002: 35).

This finding is consistent with the observations of Abel, who observed that "they (professional associations) have tended to be reactionary forces rather than professive, dedicated to preserving traditional entitlements and characteristics" (1989:131). Dacin (1997) also drew attention to the ebb and flow of macro institutional pressures. In the present case, there was substantial presence of market pressures precipitated by the foreign exchange crisis and the ICAI could have been influenced by these pressures. The rationale for the weak resistance could be thus that market pressures outweighed institutional pressures, thus limiting the ability of the ICAI's to resist the change.

Fourth, the patronage of the multinational accounting firms by the local governments and the regulators is another finding of this paper. One would have expected the Indian government to allay itself with the local accountants and the ICAI in the fight for jurisdiction in the accounting and consulting industry. While it is evident that the foreign exchange crisis in 1991 and the subsequent structural reforms arrangement led the Indian Government to authorize the



entry of the multinational accounting firms in India, the continuous and relentless patronage of the multinational accounting firms by the government as well as public sector units such as the Oil companies and the nationalized banks begs explanation. The white paper attributes this patronage to the "post-colonial hangover" effect, which basically implies that post colonial societies have a tendency to associate the work product of all western societies with higher quality and branding. Another reason forwarded by one of the authors of the white paper is that hiring the multinational accounting firms provides legitimacy to the activities of the governmental institutions, which could otherwise be questioned by any other local accounting firms. In other words, local accounting firms would be hesitant to attest certain activities of the governmental organizations due to fear of legal liability. The same activities attested by the multinational accounting firms would not attract scrutiny due to the "brand" reputation of the multinational accounting firms in the global market.

The providers of FDI, private equity and fund manager and others in the west insist on the engagement of the MAFs by the Indian corporate sector. Also the government and public sector companies and banks and financial institutions in India, which have any dealing with the global financial institutions including IMF, World Bank and other global financial and commercial institutions, are also forced to engage the MAFs in India. One important reason is that should anything were to go wrong, no one would dare to question them. Secondly, their surrogate arms were always there to attest inconvenient arrangements, which a normal Indian CA would not attest (White paper author).

Concluding Remarks

The major purpose of this paper has been to examine the interconnections of international politics and national realities affecting the globalization of the accounting profession in a weak nation state such as India. The paper has examined a particular episode in the intra-professional conflict over the jurisdiction of accounting services in India caused by the entry of the multinational accounting firms in India. The event of globalization has been often framed as involving choices between competing dualities: nationalism versus internationalism, national



sovereignty versus global citizenship, and unilateralism versus multilateralism. The evidence in this paper however raises the question of whether the interests and objectives of the agents who are constructing global markets through market expansion are compatible with broader societal interests. Specifically, the lack of level playing field for the local accounting firms in India due to the actions of international economic institutions and the multinational accounting firms threatens to undermine the development and growth of the local accounting profession in India. Internal liberalization in the accounting services market ahead of opening the market to multinational accounting firms would have prepared the local accounting firms for global competition. While internal liberalization could at least commence in the rest of the sectors of the Indian economy, such as the manufacturing sector where the tariff regime was not completely abolished and the rupee was devaluated in a phased manner, the accounting sector has not been liberalized to the present day, even ten years after the Big Four accounting firms have entrenched themselves in the national market.

These events have significance for researchers who are attempting to understand how globalization will transform professional services markets in developing economies. There are two strong themes forwarded by Hirst and Thompson (1995) which are supported by the events in this paper. First, the functions of the nation state tend to be reduced to distributing, rendering accountable and legitimizing powers of governance upwards to the international level and downwards to regional and other sub-national agencies and organized interest groups. Second, in this era of globalization, international economic institutions and multinational firms backed by major nation states play a pivotal role, while the authority and sovereignty of lesser states such as India may be continuously challenged and negotiated. Further, the role of professional associations in such a globalized environment remains invariable: the protection of the self-



interest of the most dominant members (Willmott, 1986). This is evidenced by the passage of the Chartered Accountant (Amendment) Act favoring the interests of the surrogate firms wielding a disproportionate amount of power in the ICAI. Inward looking professional associations, particularly in weaker nation-states, that due to historical reasons or by mistaken policy have failed to integrate with the dominant global system will find it increasingly difficult to survive in the long run (Caramanis, 2002).

There are many lessons to be learned from the liberalization of the Indian economy and the consequent entry of the multinational accounting firms in the Indian market for weak nationstates such as India. First, as Caramanis (2002) points out these countries provide huge opportunities to big international accounting firms for expansion and second, they are economically and politically too weak to resist significant international pressure (Garrod & McLeay, 1996). The realignment of power structures takes place on the Anglo-American free market model and therefore favors the Big Four accounting firms (Caramanis, 2002). It is therefore important for these weak nation-states to be prepared for the expansionist activities of the multinational accounting firms before these firms gain a strong foothold in the local accounting services market. Thus, national governments need to know as much about the corporate histories of major private international businesses, as they traditionally wished to know about rival states in the inter-state system (Strange, 1987). Also, professional associations in these weak nation-states have to break free from the post-colonial control of the metropolitan centers such as the UK and the USA and act to protect the interests of the local accounting actors to provide them a level playing field against the multinational accounting firms. The influence of international forces on the development of local professions has been documented in accounting academic research (Chua & Poulaos, 1993; Cooper et al., 1998; Sikka & Wilmott, 1995). The



move for international harmonisation of accounting and auditing practices has generated an increasing interest in this area.

Finally, this paper has also challenged the argument forwarded by proponents of 'globalization' that global markets are the result of inevitable and natural market forces. In fact, this analysis enforces the theory of institutional globalization (Arnold, 2002), which states that transnational institutional forces work proactively to create global markets to the detriment of the weak nation states.



References

- Annisette, M. 2000. Imperialism and the professions: the education and certification of accountants in Trinidad and Tobago. *Accounting, Organizations and Society*, 25,631-659.
- Arnold, Patricia J. 2005. Disciplining domestic regulation: the World Trade Organization and the market for professional services. *Accounting, Organizations and Society*, 30,299-330.
- Camilleri, J.A. and Falk, J. 1992. The end of sovereignty? The politics of a shrinking and fragmenting world., Edward Elgar, Aldershot.
- Campbell, D.T., and Fiske, D.W. 1959. Convergent and discriminant validation by the multitrait-multimethod matrix. *Psychological Bulletin*, 30, 116-127.
- Caramanis, C. 1999.International Accounting Firms versus indigenous auditors: Intraprofessional conflict in the Greek auditing profession. *Accounting, Organizations and Society*, 30,195-221.
- Caramanis, C. 2002. The interplay between professional groups, the state and supranational agents: Pax Americana in the age of 'globalization'. *Accounting, Organizations and Society*, 27,379-408.
- Cerra, V., Saxena, S. C., 2002. IMF staff papers, Volume 49.
- Chase-Dunn, C. 1989. Global formation: Structures of the world economy. Oxford: Blackwell.
- Chau, W.F. and Poullaos, C. 1998. The dynamics of closure amidst the construction of market, profession, empire and nationhood: An historical analysis of an Australian accounting association, 186-1903. *Accounting, Organizations and Society*, 23,155–188.
- Chua, W.F. and Poullaos, C. 1993. Rethinking the profession-state dynamic: the case of the victorian charter attempt 1885–1906. *Accounting, Organizations and Society*, 18, 91–728.
- The Chartered Accountants Action Committee For Level Playing Field. (2002). White paper on Multinational Accounting Firms operating in India, 1-141.
- Clark, G., Lakshmi, G., 2003. Controlling the risk: A case study of the Indian liquidity crisis. *Journal of International Development*, 25, 285-298.
- Cooper, D.J., Greenwood, R., Hinings, B. and Brown, J.L. 1998. Globalization and nationalism in a multinational accounting firm: The case of opening markets in Eastern Europe. *Accounting, Organizations and Society*, 23,531–548.
- Cooper, R., 1986. Economic policy in an interdependent world: Essays in world economics. MIT Press, London.



- Cox, R. 1996. Business and the state in international relations. Oxford: Westview press.
- Denzin, N.K. 1978. The research act: A theoretical introduction to sociological methods. New York: Mcgraw-Hill.
- Dicken, P. 1992. Global shift (2nd ed.)., Peter Chapman, London.
- Featherstone, M. 1990. Global culture: Nationalism, globalisation and modernity., Sage, London.
- Frieden, J. 1989. Banking on the world: The politics of international finance., Hutchinson Radius, London.
- Giddens, A. 1990. The consequences of modernity., Polity Press, Cambridge.
- Gilpin, R. 1987. The political economy of international relations., Princeton University Press, Princeton.
- Greenwood, R., Suddaby, R., Hinings, C.R. 2002. Theorizing change: The role of professional associations in the transformation of institutional fields. *Academy of Management Journal*, 45, 58-80.
- Held, D. 1991. Democracy, the nation-state and the global system. *Economy and Society*, 20(2), 138-172.
- Hirst, P. and Thomson, G. 1992. The problem of 'globalisation': international economic relations, national economic management and the formation of trading blocks. *Economy and Society*, 21,357–396.
- Hirst, P. and Thomson, G. 1995. Globalisation and the future of the nation-state. *Economy and Society*, 24, 408–442.
- Hopwood, A. 1994. Some reflections on 'the harmonisation of accounting within the EU'. *European Accounting Review* 3, 241–253.
- Jick, T. D. 1979. Triangulation in action. Administrative Science Quarterly, 601–611.
- Johnson, T.J. 1973. Imperialism and the professions: Notes on the development of professional occupations in Britain's colonies and the new states. In: Halmos, P., Editor, 1973. *Professions and social change: Sociological review monograph*, 20,186–208.
- Kapstein, E.B. 1994. Governing the global economy: International finance and the state., Harvard University Press, Cambridge, MA.
- Kennedy, P. 1993. Preparing for the twenty-first century., Random House, New York.



- Keohane, R. 1980. The theory of hegemonic stability and changes in the international system. In: Holsti, O., Siverson, R. and Alexander, G., Editors, 1980. *Changes in the international system*, Westview Press, Boulder, 131–155.
- Moizer, P. 1992. State of the art in audit market research. *European Accounting Review*, 12, 333–348.
- Ohmae, K. 1996. End of the nation state: The rise of regional economies. Special Oversees Edition., HarperCollins, London.
- Polanyi, K. 1944. The great transformation., Beacon Press, Boston.
- Puxty, A.G., Willmott, H.C., Cooper, J.D., and Lowe, T. (1987). Modes of regulation in advanced capitalism: locating accountancy in four countries. *Accounting, Organizations and Society*, 12(3), 273-291.
- Schor, J.B. 1992. Introduction. In: Banuri, T. and Schor, J.B., Editors, 1992. *Financial openness and national autonomy*, Claredon Press, Oxford, 1–14.
- Sikka, P. and Willmott, H. 1995. The power of independence: defending and extending the jurisdiction of accounting in the United Kingdom. *Accounting, Organizations and Society* 20,547–581.
- Stiglitz, J.E. 2002. Globalization and its discontents., Norton Press, New York.
- Tinker, T. 2002. Beyond the Brilovian critique: Traditional vs. organic intellectuals in critical accounting research. *Accounting and the Public Interest*, 2, 68–87.
- Willmott, H. 1986. Organising the profession: a theoretical and historical examination of the development of the major accountancy bodies in the UK. *Accounting, Organizations and Society*, 11, 555–580.



CHAPTER FOUR: STUDY THREE: ENTRY OF THE MULTINATIONAL ACCOUNTING FIRMS IN INDIA: AN INSTITUTIONAL PERSPECTIVE

Introduction

The ICAI, which does not normally get into controversies and is in fact a mechanism to discipline the profession rather than a mechanism to fight for, defend and protect the professionThe ICAI is also not structured to fight evil as it is a gentlemen's institution, not capable of handling the unequal war forced on the profession by the entry of multinational accounting firms.... (CAAC: 2002: 35).

Institutional theory represents one of the more robust sociological perspectives within organizational theory (Perrow, 2002). Dougherty pointed out that the theory contains "an excellent basis" (1994:108) for an account of change, first, by providing a convincing definition of radical(as opposed to convergent) change, and second, by describing the contextual dynamics that precipitate the need for organizational adaptation (Leblebici, Salancik, Copay, & King, 1991; Oliver, 1991). New institutionalism is, however, different from old institutional theory (Dimaggio and Powell, 1983). Under the old approach, issues of influence, coalitions, and competing values were central, along with power and informal structures (Clark, 1956; Selznick, 1949). This focus contrasts with new institutionalism with its emphasis on legitimacy, the embeddedness of organizational fields, and the centrality of classification, routines, scripts, and schema (Dimaggio & Powell, 1983; Meyer and Rowan, 1977). Organizational field is a concept which is central to institutional theory and represents an intermediate level between organization and society (Greenwood et al., forthcoming). It is actively involved in the processes by which socially constructed expectations and practices become disseminated and reproduced (Scott, 1994). While there have been a lot of studies focusing on the effects of change within a



organizational field, little is known about the way in which institutionalized practices change within an organizational field (Greenwood et al., 2002).

The present paper is a case study of a profession that underwent major change over a 15 year period. The setting is the accounting profession in India from 1990 to 2005. During this period, the Indian accounting profession underwent major changes both in terms of its market structure and jurisdiction. The change was precipitated by the foreign exchange crisis that almost turned the country bankrupt in 1991. To deal with the shortage of foreign exchange reserves leading to non payment of external debt, the government of India requested a Stand-By Arrangement from the International Monetary Fund (hereafter, IMF) in August 1991 (Clark et al., 2003). One of the stipulations of the Stand-By Arrangement was that the multinational accounting firms³, essentially the Big Four accounting firms, be allowed to provide consulting services in India (Chartered Accountants Action Committee For Level Playing Field (hereafter, CAAC), 2002). Consequently, the Reserve Bank of India issued special licenses to the multinational accounting firms authorizing them to provide consulting services in India. Significantly, the Institute of Chartered Accountants of India (hereafter, ICAI) was completely unaware of the licenses granted to the multinational accounting firms by the Reserve Bank of India to enter the Indian market (CAAC, 2002). The case study is thus an account of the change in the organizational field of the Indian accounting profession caused by the entry of the multinational accounting firms from 1990 to 2005. Specifically, I intend to understand the change in the accounting field from a social constructionist perspective using the neoinstitutional model of nonisomorphic change put forward by Greenwood et al. (2002). It is important to mention here that change cannot be treated as if it has a discrete beginning and ending (Pettigrew,

³ The terms "multinational accounting firms" and "Big Four accounting firms" refer to the same constituents and therefore are used interchangeably throughout the paper.

1987). So, the change examined in this paper in the Indian accounting profession may not sequentially follow all the stages of the model of neoinstitutional change.

This paper will examine how the entry occurred in a highly institutionalized setting and how the local profession and specifically the regulatory agency theorized and legitimated the change to themselves. The organizational field of the Indian accounting profession is a complex one, involving the regulatory agencies, the ICAI, the Indian government, the local accounting firms, the Big Four accounting firms and the Indian companies. Greenwood et al. call for "further studies in different settings to help specify the generalizability of their findings" (2002:75). This paper is a first attempt to understand change through all the stages of the institutional change model, beginning with a precipitating jolt leading into the processes of deinstitutionalization and preinstitutionalization and ending with understanding the process of theorizing the change and diffusing the change by compellingly presenting the new ideas as more appropriate than existing practices. This paper attempts to reinforce the view that professional associations are not only concerned with preserving the existing prescriptions but also play an important role in legitimating the change through a process of discourse in the field. This discourse enables the change to be resisted, debated, and eventually accepted.

The remainder of the paper is structured as follows. The next section outlines the theoretical constructs used in the paper and the nature of the Indian accounting profession, followed by a description of the methods and the data sources in the next section. Section IV presents the case study of the change in the Indian accounting profession in the past 15 years from 1990 to 2005, and the final section summarizes the conclusions of the study.



Theoretical framework

Organizational field

Scott defined an organizational field as a "community of organizations that partakes of a common meaning system and whose participants interact more frequently and fatefully with one another than with actors outside the field" (1995: 56). Organizational fields are formed through 'institutional logics' (Friedland and Alford, 1991), i.e. socially constructed rules, norms and beliefs constituting field membership, role identities and patterns of appropriate conduct. Logics conveyed through regulatory, normative and cognitive processes, shape how actors interpret reality and define the scope of socially legitimate conduct (Greenwood et al., forthcoming). The advantage of focusing upon the organizational field, rather than the individual organization is that doing so draws attention to the effects of social and technical influences on communities of organizations. Also, institutional accounts of field level processes enable us to understand the role of structuration (Greenwood et al., forthcoming). That is, organizations within a field are not only constrained by institutional structures, but, through their behaviors, act out and reproduce those structures. The result is that organizations dynamically respond to institutionalized expectations and, in so doing, amplify and elaborate them. In mature fields, therefore, organizations get progressively locked into prevailing practices.

Within a mature field, the boundaries of professional communities may be implicitly contested, but still may give the appearance of stability in the field. During these phases, practices are reproduced by regulatory and interactive processes (Greenwood et al., 2002). The appearance of stability is misleading, as fields should be seen "not as static but evolving" (Sahlin-Anderson, 1996). Fields may even "resemble institutional war" (Hoffman, 1999:352).



The Neo Institutional Model of Change

The model of nonisomorphic change distilled from the change literature by Greenwood et al. (2002) is presented in Figure 1. In this model, stage 1 refers to the dynamics that precipitate change in an organizational field. These dynamics may arise from outside the field, as destabilizing "jolts" (Meyer, 1982; Meyer, Brooks & Goes, 1990) or can also arise from endogenous sources(Seo and Creed, 2002). These jolts destabilize established practices and may take the form of social upheaval (Zucker, 1986), technological disruptions, competitive discontinuities, or regulatory change (Fox Wolfgramm, Boal & Hunt, 1998; Lounsbury, 1999; Powell, 1991). These changes precipitate (stage II) the entry of new players (Thornton, 1995), the ascendance of existing actors (Scott, Mendel, & Pollack, forthcoming), or local entrepreneurship (Dimaggio, 1988; Lawrence, 1999; Leblebici, Salancik, Copay, & King, 1991; Suddaby & Greenwood, 1999). Their effect is to disturb the socially constructed field-level consensus by introducing new ideas. Hoffman (1999) described fields as populations of

-----Insert Figure 1 here-----

organizations that, though connected and influenced by overarching logics, nevertheless subscribe to different values and beliefs. Therefore, at any moment, fields contain tensions between dominant and latent logics 'that may lie within the individual populations that inhabit the field' (Hoffman, 1999:365). Change, thus, occurs 'in the wake of triggering events that cause a reconfiguration of field membership and/or interaction pattern' (1999: 351).

The next stage (Stage III) was referred to by Tolbert and Zucker (1996) as that of preinstitutionalization, in which organizations innovate independently, seeking technically viable solutions to locally perceived problems. Initially, most researchers focused upon exogenous sources of disruption because institutions are regarded as enduring and hegemonic (Greenwood et al., forthcoming). So change was expected to arise from entrants 'transposing' ideas from one

field to another (Sewell, 1992). This explanation suggests that peripheral organizations are more likely to innovate because they are less embedded and less privileged. Organizations at the field's center, on the other hand, are more socialized, better advantaged, and thus more resistant to change (Hirsch, 1986; Galaskiewicz & Wasserman, 1989; Davis, 1991). The point to remember is that fields are never static as they are vulnerable to ideas entering from neighboring fields and to the entrepreneurial motivations of imperfectly embedded organizations (Lawrence, Hardy, & Phillips, 2002). Change is also possible in mature fields due to inner tensions, but the likelihood of change is different across fields. Some fields are more open to new ideas as their boundaries are 'permeable' (Greenwood and Hinings, 1996). Fields also differ from each other in the degree to which they have "clearly legitimated organizational templates combined with highly articulated mechanisms (the state, professional associations, regulatory agencies) for transmitting those templates to organizations..." (Greenwood and Hinings, 1996: 1029). The more complex and elaborate these templates and the stronger the mechanisms for deploying them, the more resistant the field will be to change (Greenwood et al., forthcoming).

Preinstitutionalization is followed by theorization, which refers to the conversion of ideas into compelling formats (Strang & Meyer, 1993) and is essential for granting legitimacy to the new format. Tolbert and Zucker (1996) argued that theorization involves two major tasks: specification of a general organizational failing, in order to challenge the adequacy of existing arrangements; and justification of the new organizational form as a widely appropriate solution. Prior research has examined the roles of different theorizing agents, such as management consultants (Abrahamson, 1991; Sahlin-Anderson & Engwall, 2002), the media (Davis et al., 1994; Rao et al., 2003) and professional associations (Greenwood et al., 2002). Also, cultural symbols and language have been shown to help promote or oppose existing and nascent



templates (Greenwood et al., 2002; Covaleski et al., 2003; Suddaby and Greenwood, 2005). In some cases, transorganizational structures such as professional associations increase mutual awareness and contribute towards legitimation (Greenwood et al., 2002).

Diffusion follows successful theorization and refers to the "objectification" of innovations by creating a social consensus concerning their pragmatic value (Suchman, 1995), and thus they diffuse even further (Tolbert and Zucker, 1996). There are various patterns and mechanisms by which ideas are transported within organizational communities. Lounsbury (2001) suggests that organizations of similar status respond in similar fashion, whereas Kraatz and Zazac (1996) point to the influences of local market forces. In general, groups which are not satisfied with how their interests are accommodated will favor change. The resolutions of contested values will depend upon how well groups can successfully appeal to, and draw upon, wider institutional and market forces (Greenwood et al., forthcoming).

While there have been many studies examining each of the stages of the model in different settings of change, no study has attempted to examine change in an organizational field through the lenses of all the stages of the neo institutional model of change. This paper intends to address this gap by examining the change in the Indian accounting profession over a 15 year period through a neo institutional perspective of nonisomorphic change.

Professional associations are important because they are arenas through which organizations within a professional field interact and collectively represent themselves to themselves (Greenwood et al., 2002). These interactions help define guidelines of reasonable conduct and membership. Specifically, associations enable the delineating of the domain of a profession that is what the members can do and who can do it. Second, professional associations act as the means whereby communities represent themselves to others in the field. Finally, the



professional associations can play an important role in "monitoring compliance with normatively and coercively sanctioned expectations" (Greenwood et al, 2002: 62). According to institutional theory, collective beliefs once established become taken-for-granted and then are reproduced through processes such as training and education, hiring and certification, and ceremonies of celebration. Professional associations are very active in these processes (Ruef & Scott, 1998) and therefore, "define or enforce" (Oliver, 1997: 102) such beliefs. The first two roles described above: collective representation, both internally and externally, could be in defense of the status quo or an attempt to legitimate change (Greenwood et al., 2002). This paper extends the insights offered by Greenwood et al. (2002) into these multiple roles played by professional associations by highlighting the role played by the ICAI in the change undergone by the Indian accounting profession from 1990 to 2005.

Methods

Sites and Data Sources

The field of interest in this paper is the provision of accounting services in the Indian market. This field is occupied by several organizational communities, including local accounting firms, the Big Four accounting firms, clients, and regulators such as Securities and Exchange Board of India (hereafter, SEBI) and the ICAI. The provision of accounting services in India is regulated by the ICAI and therefore these regulations can be used to demarcate the organizational community. The data sources used are archival documents and semi-structured interviews.

The main source is an extensive report about the entry of the Big Four accounting firms in the Indian market and its effects on the Indian accounting profession, in the form of a white paper published by the CAAC. The CAAC published the white paper in the "public interest to



inform the Indian industry, Indian finance sector, Indian government, Indian policy makers, Indian professionals, and also the general public about the correct facts of the multinational accounting firms (MAFs) and about the state of the Indian profession" (CAAC, 2002: 2). The white paper describes in detail the chronology of events from 1990 onwards about the entry of the Big Four accounting firms in the Indian market. Table 8 (pp.53) provides a timeline of the events leading to and consequent to the entry of the Big Four accounting firms in the Indian market.

In the late 1990s, some members of the ICAI started questioning the "west-centric model of globalization of the accounting profession in India" (CAAC, 2002:7). The process slowly evolved by extensive discussions, meetings and interactions within the Indian accounting profession. The eventual result was the birth of the CAAC at Chennai, India and the publication of the white paper. The first convention of the CAAC was held on September 1, 2002, when over four hundred Chartered Accountants (hereafter, CAs') gathered together to discuss the implications of the entry of the multinational accounting firms in the Indian market. The members of the CAAC interacted with businessmen and policy makers as well as the media all over India and found an "amazing lack of awareness about the illicit entry of the multinational accounting firms" (CAAC, 2002:7). The members of the CAAC felt that it was necessary to shed light on the causes and consequences of the entry of the multinational accounting firms on the Indian accounting profession. The CAAC was chaired by Mr. B.S. Raghavan, a retired civil servant, who was not a CA and so it was felt that the content of the white paper would not be subject to any unintended bias. The other members of the CAAC were all CAS. It is important to note here that the ICAI was neither involved in the creation of the CAAC nor in the publication of the white paper.



The research strategy adopted in this paper will begin with the white paper and will be supplemented by all the annual reports, press releases and annual president messages of the ICAI from 1990 to 2005. Other material consulted was the vision statement, which was published by the ICAI to lay out the "mission, vision and core values of the profession and repositioning the Institute and the profession of CAs' in this century" (ICAI, 2000:1). The entry of the Big Four accounting firms will be then examined in detail by consulting the materials. Appendix A provides a comprehensive list of all the authoritative archival sources used in the paper.

The data will be analyzed as follows: First, I will read the annual reports and the white paper to gain a chronology of key events such as the timing of the foreign exchange crisis, the reform package constituted by the IMF, and the entry of the Big Four accounting firms. Once the chronology of events is established, the supplemental materials such as the vision statement and the annual president messages and other ICAI reports will also be analyzed. Then, I will read the data bank; identify all the text segments⁴ that discussed the entry of the Big Four accounting firms in the Indian market. The interpretation of what constitutes discussion of the entry of the Big Four accounting firms will be liberal in order to maximize the possibility of capturing all relevant passages. The text segments will then be analyzed, using a process Gephart labeled "expansion analysis" describing it as a conceptual interpretation of the hidden meanings and features of texts" (1993:1468). . Semi-structured interviews were conducted with the key constituents in the organizational field, comprising mainly of members of the CAAC, other members of ICAI knowledgeable about the subject, and audit partners of local and Big Four accounting firms who were affected by the entry of the multinational accounting firms. The interviews were used to obtain information which may not be effectively captured in the archival

⁴ A text segment consists of two to three sentences that contain one or more key phrases. The length of the segment is determined by the need to include sufficient context to provide understanding of how the phrase or phrases is or are being used (Greenwood et al., 2002).

documents. The interviewees were asked to think about additional materials that might be available, including additional archival documents. The interviewees were also asked to discuss the contexts of the committees on which they served and of the reports, if any, they authored. Finally, the interviewees were asked to validate the chronology of events obtained from the archival documents. Appendix B provides the background of the interviewees, and the time the interviews were conducted.

The Indian Accounting Profession

In India, prior to 1949, the profession of accountancy in India was controlled and regulated by the government, which subsequently vested the regulatory power in the ICAI through an enactment in the parliament in 1949. The enactment was The Chartered Accountant Act of 1949. This act established a regulatory body for the Indian accounting profession in the form of the ICAI. The ICAI is responsible for conducting examinations, development of uniform standards of practice and the strategic development of the profession. It is also responsible for licensing, certification, training, and discipline (www.icai.org). Table 9 (pp. 56) provides a detailed timeline of events related to the birth of the ICAI to the present day.

The ICAI has its headquarters at New Delhi with five regional offices at Mumbai, Chennai, Kanpur, Kolkata, New Delhi and 98 branches spread all over the country. In addition, it has set up 11 chapters outside of India. The total membership of the ICAI is about 115,000 and over 250,000 students are pursuing the Chartered Accountancy course. Out of the total members, nearly 70% are in practice (www.icai.org). In addition to regulating the profession, the ICAI is a national standard setting body in India and the National Accounting Standards are given statutory recognition under various statutes. The Indian Accounting Standards have almost been harmonized with the International Accounting Standards. In addition, the ICAI also issues from



time to time various guidance notes, monographs, etc. for the benefit of its members on various new issues. The ICAI also issues Auditing & Assurance Standards (AAS) which codify the audit practices to be followed by the members of the ICAI, whenever an audit is carried out. The AASs' are mandatory in nature. The ICAI has issued 30 AASs' so far which meet international benchmarks and expectations.

The affairs of the Institute are managed by the Council in accordance with the provisions of the Chartered Accountant Act of 1949 and the Chartered Accountants Regulation of 1988. The Council consists of 30 members of whom 24 are elected by the members and remaining 6 are nominated by the Central Government to represent the Comptroller and Auditor General of India, Central Board of Direct Taxes, Department of Company Affairs and other stakeholders. The Council functions through 3 Standing Committees and 16 Non Standing Committees for carrying out the provisions of the Chartered Accountant Act of 1949 (www.icai.org). The decision making process in the council is structured so that all the 30 members have to consensually agree on the decision (www.icai.org). The Indian accounting firms are subject to rules and regulation made under the Chartered Accountant Act of 1949. One of the significant restraints that the ICAI places on its members is that Indian accounting firms cannot advertise, whether in India or abroad (CAAC, 2002). They cannot canvass directly or indirectly for professional assignments nor can they make presentations to prospective clients. Further, Indian accounting firms cannot have non CAS as partners nor can they have any profit sharing arrangement with non CAs' (CAAC, 2002).

The Indian accounting profession mainly consists of small and medium sized firms, the number of firms with 5 or more partners being only about 375 in 2001 out of a total of 42,339 firms. Table 10 (pp. 58) gives the distribution of firms by size in 2001 (CAAC, 2002).



The Case Study

The model of change provided in Figure 1 will be used in this section to examine the change in the Indian accounting profession from 1990 to 2005. Although I present the change in the organizational field of the Indian accounting profession sequentially as per institutional theory, it is important to reiterate that more than one of these stages could be overlapping during the process of change. In other words, change cannot be treated as though it has a discrete beginning and ending (Pettigrew, 1987), and therefore emphasis should be more on *changing* and less on *change*. For instance, it is possible that as constituents within an organizational field are engaged in innovations during the stage of preinstitutionalization, there is some degree of theorization already in progress whereby new organizational forms are gaining legitimacy through the discourses initiated by professional associations.

Precipitating Jolts and Deinstitutionalization: The Foreign Exchange Crisis (1990)

The age of liberalism, a period of relatively open global markets, ended after the Great Depression of the 1930s (Polanyi, 1944). In the post-World War II period, it was observed that many nation states instituted Keynesian economic policies and political controls over the international movement of financial capital (Schor, 1992). In the few decades after independence in 1947, India adopted a development strategy which was one of national self-sufficiency and stressed the importance of government regulation of the economy. Cerra et al. characterized it as "both inward looking and highly interventionist, consisting of import protection, complex licensing requirements, and pervasive government intervention in financial intermediation and substantial public ownership of heavy industry" (2002:3). During this time period, the trade regime in India was characterized by high nominal tariffs and pervasive non-tariff barriers, including a complex import licensing system, an "actual user" policy that restricted imports by



intermediaries, restrictions of certain exports and imports to the public sector, phased manufacturing programs that mandated progressive import substitution, and government purchase preferences for domestic producers (Cerra et al., 2002). The white paper reiterates this mindset in the Indian economy:

The socialist psyche was so much internalized in politics that in the late 1960s and early 1970s, anyone who disagreed with the socialist model was labeled as anti-poor and prorich, pro-America and even as CIA agents! Those who differed from socialist ideals were ostracized as political untouchables. With the result that, every political party was compelled to file an undertaking expressing faith in the ideology of socialism, to be registered as a political party eligible to contest elections. In fact, even the constitution of India was amended to define India as a socialist state. Even the property rights listed as unbridgeable fundamental rights were deleted and relegated to the status of mere legal rights. The effect of this is to facilitate even expropriation of property with or without adequate compensation. (CAAC, 2002: 6)

The Indian accounting profession operated in the socialist environment prevalent post independence in 1947. Prior to 1949, accounting was controlled and regulated by the Government, which eventually vested the regulatory power in the ICAI through an enactment in the parliament in the year 1949 (CAAC, 2002). The structure of the Indian accounting profession was a reflection of the structure of the Indian economy. As the partner of a local accounting firm points out, the Monopolistic and Restrictive Trade Practices Act, 1969 (MRTP Act) effectively prohibited the formation of large companies in all the sectors of the economy and so the Indian economy was comprised of small and medium-sized companies. In a similar manner, restrictions were placed on the maximum number of partners per accounting firm. Consequently, the size of an accounting firm was limited to less than 20 partners. Further, accounting firms were also prohibited from providing multi-disciplinary range of services and from entering into profit-sharing arrangements with non-members of the ICAI. In the metropolitan centers, audit work constituted the bulk of the work of medium sized accounting firms, while taxation services were



the mainstay of small sized accounting firms. The services rendered by the accounting firms included attestation services, taxation services and regulatory compliance services, which largely meant handling state regulations. The pre-dominance of audit and assurance services in the service portfolio of local accounting firms is captured in the quote below by the partner of a local accounting firm:

There was not much need for consulting services due to the nature of the market as the Indian economy was not only insulated from the global market but also had serious restraints on local market forces. On the consultancy side, which had not really grown in the absence of the market drive to the economy, handling the government, its regulations and instrumentalities for the private sector was the core part of the practice (Local firm partner).

In the 1980s and 1990s, it has been argued that most of the nation states dismantled the capital controls, bringing in the neoliberal period where once again capital is free to roam the world relatively unconstrained by national controls (Kapstein, 1994). This US agenda known as the "Washington Consensus" called not only for free trade and elimination of capital controls, but also for deregulation and economic restructuring. It was argued that IMF-imposed structural adjustment programs were used to coerce weak nation—states such as India to abandon domestic priorities in favor of open borders and investor-friendly economies (Stiglitz, 2002). In 1990, public reports of India's financial problems generated rumors of an impending foreign exchange crisis and the consequent probability of default on its borrowings from international markets and banks (Clark et al., 2003). The situation was further compounded by a variety of macroeconomic imbalances - the increase in oil prices due to the conflict in the Middle East in 1990, the reduction in remittances from Indian workers in the same region, and the reduced demand of important trading partners, as well as political uncertainty (Clark et al., 2003). To deal with the problem of shortage of foreign exchange reserves leading to non payment of external debt, the



government of India requested a Stand-By Arrangement from the IMF in August 1991. The reform package from the IMF was conditional on a structural adjustment program featuring macroeconomic stabilization and structural reforms. The reforms were mainly in the areas of industrial and import licenses, the financial sector, the tax system, and trade policy. On trade policy, benchmarks for the first review of the Stand-By Arrangement included a reduction in the level and dispersion of tariffs, a removal of quantitative restrictions on imported inputs and capital goods for export production, and elimination of public sector monopoly on imports of all items except petroleum, edible oils, and fertilizers and certain items earmarked for health and security reasons (Clark et al., 2003).

In the backdrop of these sweeping reforms, it makes it easer to analyze the process of entry of the multinational accounting firms into India. In the process of negotiating the "bailout package" with the IMF, the multinational accounting firms saw an excellent opportunity to enter the Indian market for consulting services (CAAC, 2002). One of the stipulations of the reform package from the IMF was that the multinational accounting firms be allowed to provide consulting services in India. The argument was that the multinational accounting firms were needed to facilitate the free flow of the foreign direct investments into India in the liberalized regime by the multinational corporations and investment bankers, who would only be comfortable with their own consultants as advisors about the "best ways" to invest in India (CAAC, 2002; www.icai.org; The Economic Times, 1993). Faced with a foreign exchange crisis and possibility of default on its foreign loans, the Reserve Bank of India issued special licenses to the multinational accounting firms to provide consulting services in India. In the early 1990s, the Foreign Exchange Regulation Act of 1947 (FERA) was operating in India to control the outflow of foreign exchange reserves. Under the provisions of this act, foreign firms could enter



the Indian market only after obtaining licenses issued by the Reserve Bank of India. The legal provision under which the multinational accounting firms obtained licenses to operate in India is contained under Section 29 of the FERA act, which was eventually abolished in 2002.

Post liberalization, the consulting services market provided new opportunities for the Indian accounting profession in terms of providing services to the multinational companies investing in India. However, the white paper argues that due to the entry of the multinational accounting firms, the local accounting firms could not take advantage of these opportunities as the multinational accounting firms became competitive forces in the consulting services market.

This (entry) was made possible because the MAFs managed to come to India stealthily by coining their connections in the international financial system and using the financial crisis which India faced. Thus the entire consulting services market came to be monopolized by the MAFs (CAAC, 2002: 33).

The Indian accounting profession and the ICAI were not informed about the entry of the multinational accounting firms into India, nor were they aware of the mode of entry into India. The multinational accounting firms entered India through the "floodgate of reform measures initiated by the government of India" (CAAC, 2002: 31). There was no debate about the advantages and disadvantages of the entry of the multinational accounting firms in the Indian market at any forum (CAAC, 2002; www.icai.org). No reciprocal right was conceded to the Indian accounting firms to provide accounting services in the western countries. This action is seen by the white paper as a "deliberate game plan of the west to maintain their control of the financial world" (CAAC, 2002:31). The reason for the lack of reciprocal privileges to the Indian accounting firms in the US market was the US position in the WTO that constitutional arrangement confers the right of regulation of corporate and accounting bodies on the States and the Federal Government has no authority over accounting regulation (Arnold, 2005). Thus, the



Federal Government would be violating the US constitution if it granted rights to foreign accounting firms to provide services in the USA.

Subsequently, after gaining a foothold in the consulting services market, the multinational accounting firms entered into an arrangement with a few local accounting firms turning them into "surrogates" (CAAC, 2002). The ICAI's ignorance about the entry of the Big Four accounting firms and subsequent lack of any action by the Indian accounting profession is evidenced by the reference in the white paper below:

The entry was stealthy. Even ICAI was not consulted; perhaps it knew of the decision only in newspapers. While it is true that the initial 'mistake' could probably be attributed to the RBI and the government in as much as ICAI was not even consulted before such opening up, the leadership of the profession has also failed in its part for its apparent apathy and indifference over the continued aberration, by not raising adequate objections. (CAAC, 2002: 31)

The foreign exchange reserves crisis, the resulting liberalization program instituted by the government of India, and the eventual entry of the multinational accounting firms in the Indian market are events signaling the precipitating jolt leading to the deinstitutionalization of the organizational field of the Indian accounting profession. A few elements are noteworthy. First, the foreign exchange reserves crisis that precipitated the change in the field of the Indian accounting profession arose from outside the field in the form of a destabilizing jolt (Meyer, 1982; Meyer, Brooks, & Goes, 1990). This jolt simultaneously precipitated change in the finance, tax, and accountancy fields. Second, since the jolt originated outside the field of the accounting profession, the principal constituents of the field such as the ICAI and the local accounting firms were taken by surprise by the entry of the multinational accounting firms in the Indian market. It follows that change originating from exogenous sources can disrupt the socially constructed field level consensus without the knowledge or inputs of the constituencies of the organizational field experiencing the change. Third, the entry of the Big Four accounting firms in the Indian market



and their affiliation with the local accounting firms making them into "surrogate firms" (CAAC, 2002) clearly points to the entry of new players and the ascendance of existing players (Thornton, 1995; Thornton and Ocasio, 1999; Scott, Ruek, Mendel & Caronna, 2000), thereby changing the intellectual climate of ideas in the field of the Indian accounting profession. The fact that the ICAI was not informed about the entry of the Big Four accounting firms and the local accounting firms entered into an arrangement with the new entrants, highlights the "tensions between dominant and latent logics that may lie within the individual populations (or constituencies) that inhabit the field" (Hoffman, 1999: 365). Change is therefore precipitated by triggering events that cause a reconfiguration of field membership, evidenced by the alliance between the Indian accounting firms and the Big Four accounting firms (Hoffman, 1999). The creation of "surrogate firms" out of the local Indian accounting firms by the Big Four accounting firms will be analyzed in detail in the next stage of preinstitutionalization.

Preinstitutionalization: Entry of the Big Four accounting firms in the Indian Market

The Big Four accounting firms entered the Indian market in the backdrop of the reform measures initiated by the Indian government due to the foreign exchange crisis in 1991. There was no consultation process of any kind with any stakeholder about the decision to grant licenses to the Big Four accounting firms to operate in India (CAAC, 2002). Specifically, two primary stakeholders, the ICAI and the local accounting firms were not aware of the entry of the Big Four accounting firms in the Indian market. As one of the authors of the white paper pointed out in the interview, the Big Four accounting firms entered India as consulting firms and then started providing other types of services such as globally linked taxation, non-statutory attestation work and management and proprietary audit services. However, they could not provide statutory audit services due to provisions in the Indian Companies Act of 1956, which only allowed local



accounting firms to conduct statutory audits. Therefore, the Big Four accounting firms entered into an arrangement with a few large Indian accounting firms, turning them into "surrogate firms" (CAAC, 2002:34). Each of the Big Four accounting firms has more than one surrogate firm. Most of the surrogate firms are well established large Indian accounting firms in the traditional areas of audit and assurance services (www. icai.org). The list of the surrogate accounting firms of the Big Four accounting firms is provided in Table 2 (pp.17).

This arrangement enabled the multinational accounting firms to enter the audit and assurance services market, so that the partner of the surrogate firm acted in two capacities, first as partner of the surrogate firm providing audit services and second as the director of the subsidiary of the multinational accounting firm, which was licensed to provide consulting services. Even though there were two entities, one an audit entity and another a consulting entity, both the entities shared the same premises, the same staff, the same clients and even the same website and email. With the help of the surrogate firms mentioned in the table above, the Big Four accounting firms started rendering attest and assurance services, which could only be provided by local accounting firms incorporated by CAs' licensed by the ICAI. In a report published in the "Chartered Accountant", the official publication of the ICAI, the Institute noted:

The Government should review the alternative route of entry of accounting firms in India in the name of management consulting firm, and circumvention of the law of the land taking place directly and indirectly by performing accounting services by them" (2000:16)

While the ICAI imposed strict restraints on the local accounting firms such as prohibition of canvassing, advertising their services, the Big Four accounting firms openly indulged in advertising and brand-building activities such as sponsoring cricket matches and high cost advertisement in the media through their surrogates (CAAC, 2002; www.icai.org). The Big Four accounting firms managed to advertise their services as they were not directly regulated by the



ICAI and therefore the regulations which prohibited local accounting firms from advertising or soliciting clients did not apply to the Big Four accounting firms. Further, the surrogate firms were able to convince Indian companies, which had received foreign direct investment or privately placed Foreign Institutional Investors (hereafter, FII) investment or which had entered into joint venture agreements with foreign companies, to replace their statutory and internal auditors with the surrogate firms (CAAC, 2002). The argument was that doing so would enable knowledge sharing between consulting and auditing services thereby benefiting the client. In addition to gaining an entry in the traditional audit services market, the Big Four accounting firms had one more reason to enter into an arrangement with the large Indian accounting firms. The council members of the ICAI appointed by the members of the ICAI were represented to a great degree by the partners of the surrogate firms (www.icai.org/members). Although the surrogate firms constituted less than 2% of the entire population of local accounting firms, many of the ex-presidents and current presidents were partners of the surrogate firms. For instance, the partners of two of the Big Four accounting firms currently in charge of operations in India were presidents of the council of the ICAI for the years 1998 and 1993 (www.icai.org). Therefore, the arrangement enabled the Big Four accounting firms to gain some leverage in the decision making of the ICAI.

According to the neo-institutional perspective, change is expected to arise from entrants transposing ideas from one field to another (Sewell, 1992), which explains the disruption caused by the Big Four accounting firms in the Indian accounting field. First, the Big Four accounting firms expanded the jurisdiction of the Indian accounting profession by bringing the consulting services market within the scope of the Indian accounting field. This was made possible by the creation of local accounting firms into surrogate firms providing both auditing and consulting



services. Before the entry of the multinational accounting firms, the local accounting firms primarily provided only audit and assurance services (CAAC, 2002). Second, by creating surrogate firms out of the Indian accounting firms, the Big Four accounting firms managed to enter the traditional areas of audit and assurance services, without breaking the ICAI regulation which prohibits non CAs' from providing audit and assurance services. However, neoinstitutional theory also portrays logics not as hegemonic, but as imperfectly diffused (Greenwood et al., forthcoming). This explanation suggests that peripheral organizations are more likely to innovate because they are less embedded and less privileged. Organizations at the field's center, on the other hand, are more socialized, better advantaged, and thus more resistant to change (Hirsch, 1986; Galaskiewicz & Wasserman, 1989; Davis, 1991; Leblebici et al., 1991; Kraatz and Zazac, 1996; Haveman & Rao, 1997; Palmer & Barber, 2001). In the present case however, the large well-established Indian accounting firms were the first innovators although they were in the most advantageous position before the entry of the Big Four accounting firms in the Indian market.

The explanation above would suggest that the smaller local accounting firms would be keener to accept the change and if possible, merge with the Big Four accounting firms as they were less embedded and less privileged than the large local accounting firms. However, the proposition that change originates from the periphery of a field is not absolute (e.g. Podolny, 1993; Greenwood et al., 2002; Sherer & Lee, 2002). In fact, Greenwood and Suddaby (2006) examined the introduction of a new organizational form by the Big Five accounting firms in the 1990s and concluded that markets and institutional forces unfold at different rates enabling larger, central organizations to outgrow their 'institutional moorings', allowing strategic choice. This argument explains the actions of the large accounting firms entering into an arrangement with the



Big Four accounting firms. First, the Big Four accounting firms had a competitive advantage over the local accounting firms in the consultancy services market in India due to their connection with the global financial institutions, investment bankers and rating agencies. Thus the large local accounting firms would benefit greatly from a relationship with the Big Four accounting firms. Second, the large local accounting firms were insecure about retaining their clients "in the traditional areas of audit and assurance services" (CAAC, 2002:34) due to the huge amount of foreign direct investment by multinationals in the Indian companies because these multinationals had strong links with the Big Four accounting firms. The global connections of the multinational accounting firms and the fear of losing ground in the traditional audit and assurance services market encouraged the local accounting firms to overcome their institutional "moorings" and to strategically align themselves with the multinational accounting firms.

Theorization and Diffusion: The Vision Project (1998)

The ICAI constituted a committee on Vision and Restructuring in 1998 to carry out a historic exercise of revisiting the profession and to make an attempt towards the "repositioning of the Institute and the profession of CAs' in this century" (ICAI, 2000:1). The committee published a vision statement titled "Vision for the 21st Century" in September, 2000 for public dissemination. The objective of the vision statement was very clear. It was "to formulate strategies for the success of the profession in rapidly changing economic scenario; and to give a platform for paradigm shift and to become a total business solution provider" (ICAI, 2000:2). The tone in the vision statement is clearly different from the tone in the white paper (CAAC, 2002) and both of these documents were both published at around the same time. While the white paper was clearly anti- globalization and not in favor of the entry of the Big Four



accounting firms, the vision statement considered the trend of globalization as inevitable and, therefore, implies that the entry of the Big Four accounting firms was an unavoidable fact to be faced by the Indian accounting profession. The reason could be attributed to the fact that the CAAC committee, which published the white paper, was not constituted by the ICAI. Further, the CAAC was comprised of council members, who were not associated with the Big Four accounting firms and therefore represented the interests of the local accounting firms. The committee was comprised of retired civil servants and investigative journalists, all of whom are CAS. On the other hand, the Committee on Vision and Restructuring was formed by the ICAI and mainly comprised of members associated with the surrogate firms (www.icai.org). The tone in the vision statement can be seen in the reference below:

The Indian Chartered Accountancy Profession must recognize the changes in Economy/Business environments, globalization of business and competitive pressures... and must recognize the path to success by adapting to the changes, knowledge management and acquiring skills to work with future environment influenced by technological and other changes. It must recognize the Institute's role as proactive, innovative and flexible and the need to be known as a World Class Advisor (ICAI, 2000: 1)

The vision statement appears to be a relatively clear expression of how the entry of the Big Four accounting firms and consequently the foray of accounting firms into consulting services came to be theorized by the association. The trend of globalization of accounting services was framed as "natural and inevitable" (ICAI, 2000:4). Moreover, the change was also framed as being beneficial to the Indian accounting profession. This is evident from the vision statement, which states that "The Vision describes a better vision for the profession than what existed before the Vision process. If the Vision were true today, non-accounting students would be more likely to pursue a career in the CA profession" (ICAI, 2000:7). More importantly, the change was described as within the scope of the "core values, services, and competencies" of the profession.



The core values, services and competencies are broad and provide flexibility (for the change) (ICAI, 2000:8)

This argument brings the change in normative alignment with the profession. Thus the theorization attempt so far did not mention any organizational failings as would be implied by institutional theory (Greenwood et al, 2002). On the other hand, the change was touted as a normal event to advance the progress of the Indian accounting profession.

It is worthwhile to note at this point that, while the white paper and the vision statement addressed the change in the Indian accounting profession and acknowledged the competitive pressures, an analysis of the regular publications such as The Chartered Accountant magazine and the President's messages on the ICAI website from 1995 to 2005 reveals that the ICAI continued its routine activities as before (www.icai.org). These activities were comprised primarily of clarifications of the technical rules of accounting and professional development. There were no changes in the program of precertification education to accommodate the change in the Indian accounting environment requiring the CA to "recognize the need to expand knowledge, education and experience to provide value, communicate solutions, and enhance the attractiveness of the profession" (ICAI, 2000:10). A comparison of the syllabus for the academic year 2005-2006 with the syllabus for the academic year 1994-1995 reveals no differences in the composition of the courses except for the addition of management information technology course in the year 2005, reflecting the importance of computer technology. In other words, the routines of the institute reflected the "conservative" role played by the professional associations, which is that of "monitoring prevailing institutional norms" (Greenwood et al., 2002:62). The argument within institutional theory is that once collective beliefs develop and are established, the practices associated with them are reproduced through processes such as training and education, hiring and certification (Greenwood et al., 2002). The routines of the ICAI provide evidence of



the reproduction of these established beliefs. On the other hand, the vision statement and the white paper reflected the "reforming" role of the professional associations, which is that of "collective representation in an attempt to justify the status quo", a role performed by the white paper, "or legitimate change", a role performed by the vision statement (Greenwood et al., 2002:62). In other words, the white paper was playing the conservative role of protecting the jurisdictional boundaries of the Indian accounting profession as well as the conditions of membership: that is who can practice within that jurisdiction and how. On the other hand, the vision statement was playing the reforming role of legitimating change by appealing to the "core values" of the profession.

Another theme in the vision statement was that the Indian accounting profession had to extend its focus beyond offering traditional accounting services and the need to address competition from the multinational accounting firms. These arguments were framed in a subtle manner as is evidenced by these references "A broader focus beyond "Numbers" to strategic thinking will lead to increased opportunities, professional respect, and increased rewards. The profession must address marketplace pressures from competitors in compensation packages, opportunities for rewards and advancement and who are accommodating leisure, life style and culture performances" (ICAI, 2000:4).

There are three elements worth noting here. First, the stress on focusing beyond "numbers" is an attempt to justify the actions of the "surrogate" firms offering consulting services and expanding the boundaries of the Indian accounting profession. More importantly, the argument was framed as beneficial to the entire accounting profession, not just the large accounting firms "…international arena will present even greater opportunities, not just to the large accounting firms, but to individual CAS, small firms…" (ICAI, 2000:7). This is important



as it reflects an attempt by the ICAI to justify the change in a manner more likely to appeal to majority of the local accounting firms, which were small in size. Second, the "competitors" referred to appear to be the Big Four accounting firms and the "marketplace pressures" appear to be the higher compensation packages offered by the Big Four accounting firms to hire the local CAS. In other words, the local accounting firms would have to match the increased incentives offered by the Big Four accounting firms in order to attract and retain highly qualified accountants. This position is revealing as it signals an expression by the ICAI of the inevitable presence and acceptance of the entry of the Big Four accounting firms in the Indian market. However, no mention was made of the huge differences between the resources of the Big Four accounting firms and the local accounting firms.

Third, the actions of the ICAI linking "professional respect" and "increased rewards" point to the often neglected question: Is the profession a business that pursues profit, and one that shifts its activities depending on the specific form that capital accumulation takes in different places and times? (Hanlon, 1994, 1996). Robson, Willmott, Cooper, and Puxty (1994) is an extensive exploration of the ways in which British professional bodies articulated an increasingly commercial role for their members (notably in relation to selling financial services and expanding or protecting the practice and regulatory rights of British accountants), and how this interacted with their claims to be 'professional' and acting in the public interest. Radcliffe et al. (1994) takes the analysis further by pointing out that the UK professional bodies worked hard to construct themselves, at least discursively, as enterprising organizations. The ICAI appears to attempt the same goal by emphasizing the importance of 'strategic thinking' by the local accounting firms in the changing global environment. The suggestion is that the only way for the local accounting firms to survive in the globalized environment with the entry of the



multinational accounting firms is to "adopt a structure and mindset that can rapidly adapt to the changes in the profession and the business community" (ICAI, 2005:7). The vision document states:

A core skill will be change management with willingness to abandon the old and move toward the new (ICAI, 2005:7)

Another observation is that the language used to justify the change combined the need for market positioning with the rhetoric of "providing value and service" making the debate sound more professional than businesslike. Moral legitimacy was injected into the pragmatic interests of the profession which is evident from the reference below:

Expanded Knowledge, education, experience and the seamless use of technology will create more opportunities to provide value, communicate solutions, and enhance the attractiveness of the profession. (ICAI, 2000:11)

While the entire vision statement is filled with references framing change as inevitable and the metamorphosis of the accountant in to a "total business advisor", the culmination of the endorsement for the change by the ICAI appears in stronger language in the final section of the vision statement.

The Institute's culture needs to broaden its focus from traditional services (e.g. Assurance). To win business, the Indian CAs' will need to provide value added services which would help business become more competitive. Industry specialization and detailed understanding of clients' business will underpin the ability to add value. Competitors will no longer be restricted to accounting firms. Members will compete in new market against other professional disciplines and accounting designation, where CAS will be a minority (ICAI, 2000: 35)

In other words, the vision statement exhorts the local accounting firms to expand their scope of services from the traditional areas of auditing and assurance into consulting services. Further, there is a reference to the fact that the consulting services market is primarily dominated by non CAs', thereby making CAs' a minority constituent in the new field. This reference is



particularly revealing because before the entry of the multinational accounting firms, the local accounting firms operated in a protective environment with their only competition in the form of other accounting firms. The entry of the multinational accounting firms ushered in a new competitor in the form of consulting companies.

Additional evidence of the theorization tasks carried out by the ICAI can be found in the Rules of Merger & Demerger, formulated and finalized by the ICAI on the basis of recommendations of the Study Group on Capacity Building Measures Of CA Firms (ICAI, 2005). The purpose of these rules was to regulate affiliations of two or more accounting firms providing accounting and consulting services in the Indian market. The affiliation was defined as:

Network amongst two or more firms means an arrangement to facilitate the better functioning of the affiliate member firms in the interest of the profession and not for the acquisition of any gain. Such Network shall include the Formal Network to use the collective resources such as turnover, infrastructures, manpower, and location for execution of Professional services of one or more type (ICAI, 2005).

This affiliation would also include association with an accounting entity outside India (essentially, the multinational accounting firms) such that it results directly or indirectly in a common professional economic or beneficial interest. Further, if different local accounting firms are networked with a common multinational accounting firm, then irrespective of the presence/absence of any 'affiliate' relationship between the local accounting firms, they shall be considered part of the network (ICAI, 2005). However, registration with the institute is not mandatory. These rules provide a variety of insights. First, while the stated purpose of the rules is to provide a "level playing field for the local accounting firms by pooling together their resources to enable them to compete with the surrogate firms formed by the multinational accounting firms" (ICAI, 2005: 1), as one of the authors of the white paper noted, the rules appear to be a clever attempt by the ICAI to legitimize the creation of the surrogate firms and an



endorsement of the multidisciplinary services provided by the surrogate firms. Second, the ICAI clearly attempts to provide moral legitimacy to the creation of the surrogate firms by stating that the "network" arrangement is meant to "facilitate the better functioning of the affiliate member firms in the interest of the profession and not for acquisition of any gain." Third, the provision of not requiring mandatory registration of the networks with the ICAI provides an easy escape route to the surrogate firms to avoid enforcement of compliance with the rules.

Finally, one of the most significant provisions of the rules is that if one firm of the network is the statutory auditor of an entity then the networked firms or the said firm directly or indirectly should not accept the internal audit or book-keeping or such other professional assignments, which are prohibited for the statutory audit firm. While this provision will prevent networks of local accounting firms from providing consulting services to clients to whom they provide statutory audit services, it will not have any impact on the surrogate firms formed by the multinational accounting firms. The reason is that multinational accounting firms are outside the jurisdiction of the ICAI and therefore cannot face any punitive action for their activities. This provision, therefore, places the local accounting firms at a disadvantage when competing with the multinational accounting firms.

The theorization tasks carried out by the ICAI above in the form of the vision statement and the rules of merger and demerger revolved around framing of the problem and the justification of the solution. The first task was done by framing the "change" being experienced by the Indian accounting profession as "natural and "inevitable". The Indian accounting profession has to accept and adapt to the change for the prosperity of the profession. The second task of the theorization process, which is justification of the solution, was done by combining the need for "market positioning" with the rhetoric of "adding value" and "service". The change was



described to be in alignment with the "core values" of the profession. The benefits of the change will accrue not only to the large accounting firms but also the sole proprietors and the small accounting firms. Greenwood et al. (2002) found a similar theme in case of the Canadian accounting profession where change from purely accounting services to a multidisciplinary practice concept was justified by invoking professional values. The only difference between the Canadian field and the field discussed in the present paper is that, in the former field, there was very little explicit discussion of the functionality of the change. In other words, the multidisciplinary practice concept was justified by appealing to professional values and service and not by appealing to the functional reality of market forces. In the Indian field, the change was legitimated by moral as well as functional reasoning.

Diffusion follows successful theorization (Greenwood et al., 2002). In other words, diffusion occurs if and only if, new ideas are compellingly presented as more appropriate than existing practices. This can be achieved either by nesting and aligning new ideas within prevailing normative prescriptions, thus giving them moral legitimacy(Tolbert and Zucker, 1996), and/or by asserting their functional superiority, or pragmatic legitimacy(Suchman, 1995). In the present case, the vision statement tried both methods by appealing to the local accounting firms to respond to "marketplace pressures" accompanied by framing the change to be in alignment with the "prevailing core values of the profession". Further, the rules of merger and demerger constituted by the ICAI attempted to assert the functional superiority of the 'affiliate' structure of accounting firms providing a wide range of services over the independent accounting entity providing traditional accounting and assurance services. However, as is evident from the tone of the white paper published around the same time as the vision statement, the new ideas had not yet gained social consensus concerning their pragmatic or moral value. As indicated



earlier, the reason for the same could be an unsuccessful attempt at carrying out the first step in the theorization process: framing the problem. The change was not positioned as an event to address any organizational failing as suggested by institutional theory, but as an inevitable event for the prosperity of the Indian accounting profession. The notion of "institutional war" in a field as envisaged by Hoffman (1999) is very much at play here. This fact is evident in the last section of the white paper which prescribes a "course of action" for the Indian accounting firms to counter the "threat" posed by the entry of the multinational accounting firms.

The members of the firms (local accounting firms) should not be carried away by slogans such as globalization...No further multinational accounting firms should be allowed to come in and establish business whether as consulting companies or as accounting firms....pending the outcome of the negotiation of GATS within the WTO, revoke the licenses given to these firms to operate in India (CAAC, 2002:138).

In the present setting, however, extensive changes were made to the Chartered Accountant Act, 1949 through the Chartered Accountant (Amendment) Act, 2006, which have been notified by the Central Government of India in the Gazette of India dated 23rd march, 2006 (hereafter, the Amendment Act). Some of these changes are directly relevant to the entry of the multinational accounting firms in the Indian market. The Act, for instance, has removed the restrictions on sharing of profits or fees by a member of the ICAI with certain categories of non-members, as prescribed by the ICAI from time to time, for the purpose of rendering professional services in and out of India. Further, the Act now permits a multi-disciplinary partnership, in or outside India, with certain categories of non-members, to be prescribed, from time to time by the ICAI. It is also now permitted to secure any professional business through certain categories of non-members, to be prescribed, from time to time by the ICAI. The prohibition of solicitation has been relaxed in the following circumstances: - if it occurs within the fraternity; or in the



event of responding to tenders or enquiries issued by various users. Limited advertisement (subject to guidelines to be issued by the ICAI) is also now permitted.

The new act clearly signals the objectification and institutionalization of the presence of the multinational accounting firms in the Indian market. However, the passage of the Act despite the lack of acceptance of the new ideas by all the constituents of the organizational field of the Indian accounting profession as evidenced by the white paper raises an interesting question. How can diffusion occur without successful theorization?

Hinings and Greenwood (1988) provide an explanation by pointing to the internal complexity of organizations. Building upon resource-dependence theory, they see organizations as coalitions of structurally differentiated groups. Organizations also experience contested values. The resolutions of contested values depend upon how well groups can successfully appeal to, and draw upon, wider institutional and market forces. The institutional model emphasizes the role of the institutional context, which also confers power and status upon groups, but not necessarily in a manner consistent with market forces. For instance, Greenwood et al., (2002) concluded that market conditions promoted the revenue-generating contribution of consultants within accounting firms but power remained with accountants because of regulations prohibiting non-accountants from controlling accounting firms. In the present setting, however, the passage of the act allowing non-members, essentially members of the multinational accounting firms, to enter into profit sharing arrangements with members of the ICAI, signals the transfer of power within the surrogate firms from members of the local accounting arm of the surrogate firm to the partners or members of the multinational accounting firms. As one of the authors of the white paper points out:

Now the Big Four no longer have to hide behind the guise of the local accounting firms and can openly provide audit and assurance services as long as they have one local CA as a



partner working for them. The balance of power is bound to shift (in favor of the multinational accounting firms) (White paper author)

This suggests that when the mechanisms (in this case, the ICAI and the regulatory agencies) for protecting the institutional templates within an organizational field are weak, market forces can outweigh institutional forces resulting in new organizational structures as dictated by the market forces.

Discussion and Conclusion

Role of professional associations in the process of change

Professional associations have been argued to perform three roles: the reproduction of prevailing logics through repetitive routines, hosting intraprofessional discourse, and representing the profession to outsiders (Greenwood et al, 2002). The first role has been argued to be a more conservative role, whereas the last two roles may not have conservative outcomes. For instance in the current setting, the ICAI's routines reinforced and reproduced the prevailing logic of action (first role), while the same association hosted and narrated the debate over which the creation of surrogate firms by the multinational accounting firms came to be legitimated and consulting services and activities were included within the jurisdiction of accounting.

In the current setting, similar to Greenwood et al. (2002), the ICAI did not initiate the change. The change was precipitated by the jurisdictional and organizational movements of the profession's largest firms, specifically the large local accounting firms which formed a surrogate arrangement with the multinational accounting firms. The ICAI merely responded to these actions of the firms. This pattern is consistent with the observations of Greenwood et al. (2002) who argued that the large firms had three reasons to act as institutional entrepreneurs. First, these firms were less captured by the prevailing routines. Second, they were attuned to the emerging



opportunities presented by the entry of the multinational accounting firms with their strong connections in the global financial system. Third, they wielded a disproportionate amount of power in the profession through representation in the ICAI and therefore were better equipped to resist traditional practices. The important role played by the ICAI was to frame the entrepreneurial movement in terms that would nullify any opposition or resistance by the other constituents within the organizational field. In the present setting however, there was sharp resistance as is evidenced by the tone of the white paper and the opposition to the entry of the multinational accounting firms in the Indian market. Although the ICAI attempted to provide moral and functional reasons to legitimate the entry of the multinational accounting firms and the evolution of a "total business provider" replacing the traditional CA, it would be naïve to infer from the evidence in this paper that the legitimating activities of the ICAI succeeded in completely nullifying the opposition to the change. So the question arises: Why did the ICAI seek to legitimize the change in spite of the strong opposition to the change from various constituents of the organizational field?

It has been argued that increasing pressure of market forces over a long period of time, in this case over eight years, can alter the balance between institutional forces and market forces in favor of the latter, thus forcing the ICAI to play its reforming role and legitimate the change (Greenwood et al., forthcoming). However, the evidence in this paper suggests two more reasons. First, as explained earlier, the larger firms which were involved in the surrogate arrangement had the political resources to influence the actions of the ICAI. Second, the ICAI was placed in a position whereby its own legitimacy was endangered. The reason is that the multinational accounting firms who created the surrogate firms and expanded the boundaries of the Indian accounting profession were not within the jurisdiction of the ICAI. So the ICAI could not resist



or oppose any entrepreneurial movements of the multinational accounting firms in a realistic fashion. Consequently, the ICAI had no option but to endorse the presence of the multinational accounting firms and consequently the activities of the surrogate firms. In other words, the role of the professional association may be multi-faceted altering with the circumstances of the marketplace. Once its own legitimacy was challenged, the reproductive role gave way to the reforming role thereby enabling the legitimation of the change.

Finally, it has been argued that change follows conflict of interests (Abbott, 1988; Hoffman, 1999). In the present setting, there were sharp differences of interests between professional subcommunities, and it was the larger and the more powerful subcommunity that desired change. In the Canadian setting, Greenwood et al, (2002) concluded that despite the lack of conflict of interests, the profession sought a normative endorsement of the change. This explanation would suggest that in the presence of sharply contested interests as observed in the current setting, the ICAI should have appealed more strongly to moral reasoning rather than market forces to justify the change. However, as the evidence in the paper suggests, while the ICAI did frame the change to be within the "core values" of the profession, it also appealed to the need to "recognize the changes in the global environment, globalization of business and competitive pressures" (ICAI, 2000:1) while justifying the change in the theorization process. One possible explanation could be the "permeability" or "plasticity" (Fox-Wolfgram et al., 1998) of professional boundaries. Professional communities may respond differently to the movements of constituent subcommunities. The legal profession has, for instance, been argued to be less open than accountancy to new ideas and law societies have been pronouncing against multidisciplinary practices (Greenwood et al., 2002). The reason forwarded is that the accounting profession exhibits greater plasticity in its identity than does the legal profession. However, the



preceding discussion suggests that accounting professions in different settings may also exhibit different plasticities. For instance, the Indian accounting profession may respond differently to change despite sharply contested interests as opposed to the Canadian accounting profession. Further research is needed to uncover the determinants of the plasticities of different professions or the same profession in different settings. Another explanation could be the role played by professional associations in the commercialization of the accounting profession. For example, it would be useful to examine whether claims to 'gentlemanly' and professional values might be a strategy to be commercial. Appeals to such forms of social and cultural capital can be an effective strategy for professionals in developing their business and being profitable (Cooper, Hinings, Greenwood, & Brown, 1996).

Theorization and Diffusion

Legitimation, which is the central task of theorization, has been argued to take different forms in professionalized setting as opposed to commercial settings (Greenwood et al., 2002). In a highly professionalized setting, it has been argued that commercial pressures may precipitate institutional entrepreneurship, but diffusion requires a normative justification. On the other hand, in commercial settings, it has been argued that change often spreads through the mimetic mechanism (Dimaggio & Powell, 1983; Strang & Soule, 1998). For instance, Haveman (1993) showed that savings and loans firms entered new market niches in increasing numbers once those niches had been occupied by large reputedly successful corporations. Similarly, Haunschild and Miner (1997) showed that firms' choices of investment bankers were affected by earlier choices made by large successful firms. These studies imply that mimicry occurs because a course of action is legitimated by its use of others. The basis of legitimation is primarily economic. In the present professionalized setting, however, one can see that the mimetic mechanism is more



apparent. While the ICAI did try to frame the change within the values embedded in traditional beliefs, it primarily appealed to the "market potential and advantages" of the change. Also, once the surrogate firms formed by the multinational accounting firms and large local accounting firms were formed, the ICAI constituted the Rules of Merger and Demerger (ICAI, 2005) exhorting the local accounting firms to form 'networks' of firms to provide services to clients. These rules clearly seek to mimic the surrogate arrangement innovated by the large well-established firms in the Indian accounting profession. This leads us to believe that the proposition that legitimation in professional setting has to be a function of professional appropriateness alone, seems to be an oversimplification.

Further, it has also been argued that diffusion follows successful theorization. However, as the evidence in the paper suggests, the ICAI's attempts at theorization cannot be termed as successful as evidenced by the tone in the white paper and its open call for the withdrawal of the licenses granted to the multinational accounting firms to operate in India. The passage of the Chartered Accountant (Amendment) Act, which by recognizing partnerships between members (local accounting firms) and non-members (multinational accounting firms) clearly points to the institutionalization of the presence of the multinational accounting firms in India. One possible explanation, as outlined earlier, is that organizations are complex entities with highly contested values. The resolutions of contested values depend upon how well groups can successfully appeal to, and draw upon, wider institutional and market forces and differs from field to field (Hinings and Greenwood, 1996). Fields vary in the degree to which they have clearly legitimated organizational templates combined with highly articulated mechanisms (the state, professional associations, regulatory agencies). In fields where the mechanisms for the deployment of the templates are weak, theorizing the change and subsequent diffusion of new ideas might be done



more easily than in fields with stronger mechanisms. Further research is needed to draw out the complex ways through which new ideas get institutionalized and objectified in similar settings.

In this paper, I have examined the change in the Indian setting through the lenses of all the stages of the model of non-isomorphic change in institutional theory. Further, I have attempted to understand the diverse roles played by professional associations during the process of change. Specifically, I have examined the role of professional associations in the theorization process against the backdrop of sharply contested interests in the organizational field of the accounting profession. Finally, this paper has demonstrated the diffusion of new ideas in an accounting setting despite unsuccessful execution of the theorization process. Future research will be aimed at examining each of the stages in the model in a more detailed manner in the current setting.



References

- Abrahamson, E. 1991. Managerial fads and fashions: The diffusion and rejection of innovations. *Academy of Management Review*, 16 (3): 586-612.
- The Chartered Accountants Action Committee For Level Playing Field. 2002. White paper on Multinational Accounting Firms operating in India: 1-141.
- Cerra, V., Saxena, S. C., 2002. IMF staff papers, Volume 49.
- Clark, B. 1956. Organizational Adaptation and precarious values. *American Sociological Review*, 21:7
- Clark, G., Lakshmi, G., 2003. Controlling the risk: A case study of the Indian liquidity crisis. Journal of International Development, 25:285-298.
- Covaleski, M., Dirsmith, M.W., Rittenberg, L. 2003. Jurisdictional disputes over professional work: The institutionalization of the global knowledge expert. *Accounting Organizations and Society*, 28 (4): 325-355.
- Davis, G.R. 1991. Agents without principle? The spread of the poison pill through the intercorporate network. *Administrative Science Quarterly*, 36: 583-613.
- Davis, G.F., Diekmann, K.A., Tinsley, C.H. 1994. The decline and fall of the conglomerate firm in the 1980s: The deinstitutionalization of an organizational form. *American Sociological Review*, 59 (4): 547-570.
- DiMaggio, P.J. 1988. Interest and agency in institutional theory. In L.G. Zucker (ed.), *Institutional Patterns and Organizations: Culture and Environment*. Cambridge, MA: Ballinger.
- DiMaggio, P.J., Powell, W.W. 1983. The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*, 48: 147-160.
- Dougherty, D. 1994. Commentary. In P. Srivastava, A. Huff, & J. Dutton (Eds.), *Advances in strategic management*, vol. 10: 107-112. Greenwich, CT: JAI Press.
- Fox-Wolgram, S. J., Boal, K. B., & Hunt, J.G. 1998. Organizational adaptation to institutional change: A comparative study of first-order change in prospector and defender banks. *Administrative Science Quarterly*, 43: 87-126
- Friedland, R., Alford, R.R. 1991. Bringing society back in: Symbols, practices and institutional contradictions. In W.W. Powell & P.J. DiMaggio (eds.), *The New Institutionalism in Organizational Analysis*. Chicago, IL: University of Chicago Press.



- Galaskiewicz, J., Wasserman, S. 1989. Mimetic processes within an interorganizational field. *Administrative Science Quarterly*, 34 (3): 454-480.
- Gephart, R. P. 1993. The textual approach: Risk and blame in disaster Sensemaking. *Academy of Management Journal*, 36: 1465-1515.
- Greenwood, R., Hinings, C.R. 1996. Understanding radical organizational change: Bringing together the old and the new institutionalism. *Academy of Management Review*, 21 (4): 1022-1055.
- Greenwood, R., Suddaby, R. 2006. Institutional entrepreneurship in mature fields: The Big Five accounting firms. *Academy of Management Journal*, forthcoming.
- Greenwood, R., Suddaby, R., Hinings, C.R. 2002. Theorizing change: The role of professional associations in the transformation of institutional fields. *Academy of Management Journal*, 45 (1): 58-80.
- Green, R., Hinings, C.R. forthcoming. Radical Organizational Change. *Handbook of Organization Studies*
- Hinings, C.R. 1979. Continuities in the Study of Organizations. In, C. Lammers & D.J. Hickson (eds.), *Organizations Alike and Unlike*. London, Boston: Routledge and Kegan Paul.
- Hirsch, P.M. 1986. From ambushes to golden parachutes: Corporate takeovers as an instance of cultural framing and institutional integration. *American Journal of Sociology*, 91 (4): 800-37.
- Hoffman, A.J. 1999. Institutional evolution and change: Environmentalism and the US chemical industry. *Academy of Management Journal*, 42 (4): 351-371.
- Hoffman, A.J., Ventresca, M.J. 1999. The institutional framing of policy debates. *American Behavioral Scientist*, 42 (8): 1368-1392.
- The Institute of Chartered Accountants of India.2000. The Committee on Vision and Restructuring. 2000. *The Vision statement*. ICAI.
- Kraatz, M.S., Zajac, E.J. 1996. Exploring the limits of the new institutionalism: The causes and consequences of illegitimate organizational change. *American Sociological Review*, 61 (5): 812-836.
- Lawrence, T., Hardy, C., Phillips, N. 2002. Institutional effects of interorganizational collaboration: The emergence of proto-institutions. *Academy of Management Journal*, 45:281-291.



- Leblebici, H., Salancik, G., Copay, A., King, T. 1991. Institutional change and the transformation of the US radio broadcasting industry. *Administrative Science Quarterly*, 36: 333-363.
- Levinthal, D.A., March, J.G. 1993. The myopia of learning. *Strategic Management Journal*, 14 (Special Issue): 95-112.
- Lounsbury, M. 2001. Institutional sources of practice variation: Staffing college and university recycling programs. *Administrative Science Quarterly*, 46 (1): 29-56.
- Lounsbury, M. 1999. From heresy to dogma: An institutional history of corporate environmentalism. *Administrative Science Quarterly*, 44 (1): 193-196.
- Meyer, A.D. 1982. Adapting to environmental jolts. *Administrative Science Quarterly*, 27 (4): 515-537.
- Meyer, A.D., Brooks, G.R., Goes, J.B. 1990. Environmental jolts and industry revolutions: Organizational responses to discontinuous change. *Strategic Management Journal*, 11 (Special Issue): 93-110.
- Meyer, J.W., Rowan, B. 1977. Institutionalized organizations: Formal structure as myth and ceremony. *American Journal of Sociology*, 83 (2): 440-463.
- Miles, R.E., Snow, C.C. 1978. *Organizational Strategy, Structure, and Process*. McGraw-Hill Series in Management. New York: McGraw-Hill.
- Ocasio, W., Kim, H. 1999. The circulation of corporate control: Selection of functional backgrounds of new CEOs in large US manufacturing firms, 1981-1992. *Administrative Science Quarterly*, 44: 532-62.
- Oliver, C. 1991. The antecedents of deinstitutionalization. *Organization Studies*, 13 (4): 563-588.
- Oliver, C. 1997. The influence of institutional and task environment relationships on organizational performance: The Canadian construction industry. *Journal of Management Studies*, 34: 99-124.
- Palmer, D., Barber, M. 2001. Challengers, elites and owning families: A social class theory of corporate acquisitions in the 1960s. *Administrative Science Quarterly*, 46: 87-120.
- Palmer, D., Jennings, P.D., Zhou, X. 1993. Late adoption of the multidivisional form by large US corporations: Institutional, political and economic accounts. *Administrative Science Quarterly*, 38 (1): 100-132.
- Perrow, C. 2002. Organizing America Wealth, Power and the Origins of Corporate Capitalism. Princeton: Princeton University Press.



- Pettigrew, A.M. 1987. Context and action in the transformation of the firm. *Journal of Management Studies*, 24 (6): 649-670.
- Pettigrew, A.M. 1985. The Awakening Giant: Continuity and Change in Imperial Chemical Industries. Oxford, UK: Blackwell.
- Pettigrew, A.M. 1983. Patterns of managerial response as organizations move from rich to poor environments. *Educational Management Administration*, 2: 104-114.
- Pettigrew, A.M., Fenton, E.M. 2000. Complexities and dualities in innovative forms of organizing. In, A.M. Pettigrew & E.M. Fenton (eds.), *The Innovating Organization*. London, UK: Sage Publications.
- Pettigrew, A., Ferlie, E., McKee, L. 1992. Shaping Strategic Change. London: Sage.
- Pettigrew, A.M., Whipp, R. 1991. *Managing Change for Competitive Success*. Oxford: Blackwell.
- Podolny, J.M. 1993. A status-based model of market competition. *American Journal of Sociology*, 98: 829-872.
- Powell, W. W. 1991. Expanding the scope of institutional analysis. In W. W. Powell and P. J. Dimaggio (Eds.), *The new institutionalism in organizational analysis*. Chicago: University of Chicago press.
- Powell, W.W., DiMaggio, P.J. 1991. *The New Institutionalism in Organizational Analysis*. Chicago, IL: University of Chicago Press.
- Pugh, D.S., Hickson, D.J., Hinings, C.R. 1969. The context of organization structures. *Administrative Science Quarterly*, 14: 91-114.
- Quinn, R.E., Cameron, K.S. 1983. Organizational life cycles and shifting criteria of effectiveness: Some preliminary evidence. *Management Science*, 29: 33-51.
- Rao, H., Monin, P., Durand, R. 2003. Institutional change in Toque Ville: Nouvelle cuisine as an identity movement in French gastronomy. *American Journal of Sociology*, 108 (4): 795-843.
- Ruef, M. 2000. The emergence of organizational forms: A community ecology approach. *American Journal of Sociology*, 106 (3): 658-714.
- Ruef, M., & Scott, W. R. 1998. A multidimensional model of organizational legitimacy: Hospital survival in changing institutional environments. *Administrative Science Quarterly*, 43: 877-904.



- Sahlin-Andersson, K. 1996. Imitating by editing success: The construction of organizational fields. *In B. Czarniawska & G. Sevon (Eds.), Translating organizational change*: 69-92. Berlin: de Gruyter.
- Sahlin-Anderson, K., Engwall, L. (eds.) 2002. The Expansion of Management Knowledge: Carriers, Flows, and Sources. Stanford, CA: Stanford Business Books.
- Salancik, G.R. 1995. Wanted: A good network theory of organization. *Administrative Science Quarterly*, 40: 345-349.
- Scott, W. R. 1994. Conceptualising organizational fields: Linking organizations and societal systems. In H. U. Derlien, U. Gerhardt, & F. W. Scharpf (Eds.), *Systemrationalitat und partialinteresse* [Systems rationality and partial interests]: 203-221. Baden Baden, Germany: Nomos Verlagsgesellschaft.
- Scott, W.R. 1995. Institutions and Organizations. Thousand Oaks, CA: Sage.
- Scott, W.R., Ruef, M., Mendel, P.J., Caronna, C.A. 2000. *Institutional Change and Healthcare Organizations: From Professional Dominance to Managed Care*. Chicago, IL: University of Chicago Press.
- Scott, W. R., Mendel, P., & Pollack, S. Forthcoming. Environments and fields: Studying the evolution of a field of medical care organizations. In W. W. Powell and D. Jones (Eds.), *How institutions change*. Chicago: University of Chicago Press
- Selznick, P. 1949. TVA and the Grassroots. Berkeley, CA: University of California Press.
- Sewell, W.H., Jr. 1992. A theory of structure: Duality, agency, and transformation. *American Journal of Sociology*, 98: 1-29.
- Seo, M-G., Creed, W.E.D. 2002. Institutional contradictions, praxis, and institutional change: A dialectic perspective. *Academy of Management Review*, 27 (2): 222-247.
- Sherer, P.D., Lee, K. 2002. Institutional change in large law firms: A resource dependency and institutional perspective. *Academy of Management Journal*, 45: 102-119.
- Strang, D., Meyer, J.W. 1993. Institutional conditions for diffusion. *Theory and Society*, 22: 487-511.
- Strang, D., Meyer, J. W. 1993. Institutional conditions for diffusion. Theory and Society, 22: 487-511.
- Suddaby, R., Greenwood, R. 1999. The colonization of knowledge: Product commodification as a dynamic of change in professional service firms. Paper presented at the SCANCOR Conference, Stanford University, Stanford, CA.



- Suddaby, R., Greenwood, R. 2005. Rhetorical strategies of legitimacy. *Administrative Science Quarterly*, March.
- Thornton, P.H. 2004. Markets from Culture. Stanford, CA: Stanford University Press.
- Thornton, P.H. 1995. Accounting for acquisition waves: Evidence from the US college publishing industry. In W.R. Scott & S. Christensen (eds.), *The Institutional Construction of Organizations: International and Longitudinal Studies*. Thousand Oaks, CA: Sage.
- Thornton, P.H., Ocasio, W. 1999. Institutional logics and the historical contingency of power in organizations: Executive succession in the higher education publishing industry, 1958-1990. *American Journal of Sociology*, 105 (3): 801-843.
- Tolbert, P.S., Zucker, L.G. 1996. The institutionalization of institutional theory. In S.R. Clegg, C. Hardy & W.R. Nord (eds.), *Handbook of Organizational Studies*. Thousand Oaks, CA: Sage.
- Tolbert, P.S., Zucker, L.G. 1983. Institutional sources of change in the formal structure of organizations: The diffusion of civil service reform, 1880-1935. *Administrative Science Quarterly*, 30: 22-39.
- Zucker, L. G. 1986. Production of trust: Institutional sources of economic structure, 1840-1920. In B. M. Staw & L. L. Cummings (Eds.), *Research in organizational behavior*, vol. 8: 53-112. Greenwich, CT: JAI Press.



II. Deinstitutionalization III. Emergence of new **Preinstitutionalization** I. Precipitating Jolts players Independent \Box Ascendance of Social innovation actors Technological Technical viability Institutional Regulatory paramount entrepreneurship IV. Theorization VI. Reinstitutionalization Specification of V. Diffusion Cognitive legitimacy general Increasing \Box organizational objectification failing Pragmatic Justification of legitimacy abstract possible solution Moral and/or pragmatic legitimacy

Figure 1:Neo-Institutional-Model-of-change

APPENDIX A: LIST OF AUTHORATATIVE SOURCES



Document	Year	Source	
White Paper on Multinational	2002	The Chartered Accountants	
Accounting Firms operating in		Action Committee For Level	
India		Playing Field	
Vision Document	2005	The Institute Of Chartered	
		Accountants Of India (ICAI)	
ICAI President Messages	1990-2005	ICAI website	
ICAI Annual Reports	1990-2005	ICAI website	
Rules of Network and Merger-	2004	ICAI website	
Demerger amongst the firms			
registered with the ICAI			
The Chartered Accountant	2006	ICAI website	
(Amendment) Act			
World Trade Organization	2002	The Chartered Accountant	
GATS Regime: Implications		Magazine	
For the Accountancy Sector			
ICAI Patrika	2005-2006	ICAI website	
Student Newsletter (ICAI)	1995-2006	ICAI (New Delhi)	
Press Releases(ICAI)	1995-2006	ICAI(New Delhi)	



APPENDIX B: INTERVIEW DETAILS



Date of	Interviewee	Country	Occupation	Work
Interview		•	•	Experience
July 2006	201	India	Head of Audit	More than 10
-			Function in a Big	years of audit
			Four Firm in	experience in the
			India	Indian market
				and director of a
				Big Four
				surrogate firm
July 2006	202	India	Head of Audit	More than 8
			Function in a Big	years of
			Four Firm in	experience in
			India	audit and director
				of a Big Four
				surrogate firm
July 2006	203	India	Senior Audit	Five years of
			Manager in a Big	audit experience
			Four Firm in	with a medium
			India	sized local
9 1 2006	20.4	T 11		accounting firm
September 2006	204	India	Author of the	A Chartered
			White Paper	Accountant and
				an investigative
C t 1 2006	205	India	Author of the	journalist Retired
September 2006	205	maia		
Santambar 2006	206	India	White Paper Author of the	Bureaucrat
September 2006	200	Illula		Chartered
Santambar 2006	207	India	White Paper Author of the	Accountant Chartered
September 2006	207	Illula	White Paper	Accountant
October 2006	208	India	Partner of a local	More than ten
October 2000	208	Illula	accounting firm	years of audit
			in India	experience in
			III IIIUIa	India
October 2006	209	India	Partner of a local	More than fifteen
200001 2000	207	IIIGIU	accounting firm	years of audit
			in India	experience in
			111111111111111111111111111111111111111	India

